EXTERNAL CERVICAL ROOT RESORPTION: DETERMINANTS & TREATMENT OUTCOMES

by

Eleni Irinakis

D.T., Athens University of Applied Sciences, 2008

D.D.S., Aristotle University of Thessaloniki, 2014

A THESIS SUBMITTED IN PARTIAL FULFILLMENT

OF THE REQUIREMENTS FOR THE DEGREE OF

MASTER OF SCIENCE

in

THE FACULTY OF GRADUATE AND POSTDOCTORAL STUDIES

(Craniofacial Science)

THE UNIVERSITY OF BRITISH COLUMBIA

(Vancouver)

October 2018

©Eleni Irinakis, 2018

The following individuals certify that they have read, and recommend to the Faculty of Graduate and Postdoctoral Studies for acceptance, the thesis entitled:

External Cervical Root Resorption: Determinants and Treatment Outcomes

submitted by <u>Eleni Irinakis</u> in partial fulfillment of the requirements for the degree of <u>Master of Science</u>

in the Faculty of Graduate and Postdoctoral Studies (Craniofacial Science)

Examining Committee:

<u>Dr. Markus Haapasalo</u>	Co-supervisor
Dr. Jolanta Aleksejuniene	Co-supervisor
Dr. Ya Shen	Committee Member
Dr. Ahmed Hieawy	Committee Member
Dr. Nesrine Mostafa	External Examiner

Abstract

Objectives: The objectives of this study were to assess if any systemic condition could be a potential predisposing factor for external cervical root resorption (ECRR), and to assess the long-term ECRR treatment outcome and its determinants.

Methods: This study contains data from 76 patients (98 teeth) diagnosed with ECRR at the UBC Graduate Endodontics clinic, from 2008 to 2018. Data regarding the medical and dental history were retrospectively collected from the charts of the ECRR group and an equivalent group of patients without ECRR (control group). Subsequently, the ECRR patients were approached for a follow-up appointment, during which a clinical examination was conducted and intraoral photos were taken. Periapical radiographs, and CBCT if indicated, were taken for the radiographic evaluation. Chi Square test or Fisher's Exact test were used for statistical comparisons at two levels, patient-based and tooth-based level. The Kaplan Meier curves method was used in order to evaluate the overall ECRR survival/failure rates, and how various treatment-related and local predisposing determinants were associated with the ECRR treatment outcome.

Results: Overall, 67 patients were evaluated. The mean follow-up was 3.9 years with the minimum follow-up being one year. The majority of the patients were older than 40 years old (72.4%). The most frequently affected teeth were the maxillary anteriors (31.7%) with the most common diagnosis being Class 2 (38.8%). Half of the cases survived for eight years. Twenty-four teeth failed (i.e. 19 extracted, 5 not functional). The only influencing factor that proved to be statistically significant among the systemic conditions was diabetes, and it was more frequently present in the ECRR group than in the control group. Determinants with statistically significant influence were: the root canal treatment (RCT) and the resorption repair combined with RCT as

iii

local determinants; and the tooth location and the Heithersay classification as treatment-related determinants.

Conclusions: Diabetes may be a potential systemic predisposing factor for ECRR. RCT and the ECRR repair combined with RCT are associated with lower failure rates. Higher failure rates are associated with posterior teeth and higher classes in the Heithersay classification.

Lay summary

External cervical root resorption (ECRR) is a poorly investigated, dynamic process. The key goals of the present study were to assess if any systemic conditions predispose ECRR, and to examine the ECRR treatment outcomes and its determinants.

The study's contribution was to provide a new clinical research approach to this challenging topic. More specifically, there are no previous cohort studies about the association between systemic determinants and ECRR. The present study suggests that diabetes may be a risk factor for ECRR. This indication is of significance because it may have an impact in early diagnosis of ECRR. In addition, cohort studies have not been conducted regarding how various ECRR treatment-related and local determinants may be associated with the treatment outcome. Our findings are of clinical importance because they can provide indications for better treatment planning, and they also highlight the significance of early ECRR diagnosis.

Preface

The present study was designed and completed by the MSc candidate, Eleni Irinakis, under the supervision and guidance of Dr. Markus Haapasalo and Dr. Jolanta Aleksejuniene. The relative contribution of the collaborators in this study was: Dr. Eleni Irinakis 80%, Dr. Jolanta Aleksejuniene 10% and Dr. Markus Haapasalo 10%. The project was approved by the UBC Behavior Research Ethics Board (H17-00885).

Table of Contents

Abstract	iii
Lay summa	ryv
Preface	vi
Table of Co	ntentsvii
List of Table	esx
List of Figur	esxi
List of Abbr	eviationsxiv
Acknowled	gementsxv
Dedication.	xvi
Chapter 1: I	literature review1
1.1	Introduction1
1.2	Classification of root resorption2
1.3	Histology of root resorption5
1.3.1	Physiological root resorption6
1.3.2	Pathological root resorption7
1.4	Classification of external cervical root resorption11
1.5	Mechanism of action of external cervical root resorption15
1.5.1	External cervical root resorption in vital teeth15
1.5.2	External cervical root resorption in endodontically treated teeth 17 vii

1.6		ology and predisposing factors	.17
1.0	5.1	Predisposing factors based on cohort studies	. 18
1.6	5.2	Predisposing factors based on case and case series reports	. 20
1.6	i.3	Idiopathic external cervical root resorption	. 28
1.6	5.4	Multifactorial etiology	. 29
1.7	Di	agnosis	.29
1.7	'.1	Clinical findings	. 30
1.7	.2	Radiographic findings and tools	. 31
1.8	Tr	eatment management	.34
1.8	8.1	External cervical root resorption defects Class 1 and 2	. 36
1.8	8.2	External cervical root resorption defects Class 3 and 4	. 36
1.8	.3	Treatment considerations and prognosis	. 38
1.9	Tr	eatment outcome	.39
Chapter 2	: Hvo	otheses and objectives	40
•			
Chapter 3	: Ma	terials and methods	41
Chapter 3 3.1	: Ma i Sai	terials and methods	41 .41
Chapter 3 3.1 3.2	: Ma Sai Da	terials and methods mple selection ta collection	41 .41 .48
Chapter 3 3.1 3.2 3.3	: Ma Sa Da Da	terials and methods mple selection ta collection ta analyses	41 .41 .48 .53
hapter 3 3.1 3.2 3.3 hapter 4	Sai Da Da	terials and methods mple selection ta collection ta analyses	41 .41 .48 .53 54
Chapter 3 3.1 3.2 3.3 Chapter 4 4.1	Sai Da Da : Res Pa	terials and methods mple selection ta collection ta analyses ults tient level of analysis	41 .41 .53 54 .54
hapter 3 3.1 3.2 3.3 hapter 4 4.1 4.2	: Ma Sai Da Da : Res Pa To	terials and methods	41 .41 .53 54 .54
Chapter 3 3.1 3.2 3.3 Chapter 4 4.1 4.2 Chapter 5	: Ma Sau Da Da : Res Pa To	terials and methods mple selection ta collection ta analyses tianalyses ults tient level of analysis oth level of analysis	 41 .41 .48 .53 54 .61 78
Chapter 3 3.1 3.2 3.3 Chapter 4 4.1 4.2 Chapter 5 5.1	: Ma Sai Da Da : Res Pa To : Dis	terials and methods	 41 .41 .48 .53 54 .61 78 .78

Referenc	es	87
Chapter (5: Conclusions	86
5.4	Future directions	85
5.3	Limitations	84
5.2	Novelty and significance	83

List of Tables

Table 1. Potential determinants: presence of systemic conditions	.57
Table 2. Treatment related local determinants in association with failure rates	67
Table 3. Local determinants in association with failure rates	68

List of Figures

Figure 1. Terminology and classification of root resorption according to the American	
Association of Endodontists	4
Figure 2. Histological findings	10
Figure 3. Illustration of the Heithersay classification of external cervical root resorption	12
Figure 4. Illustration of a novel 3D classification	14
Figure 5. Periapical radiograph and clinical photo of tooth with pink spot at the cervical	
area	30
Figure 6. A print screen of a patient's chart in Romexis	42
Figure 7. Excel sheet illustrates the organization of the information given about the key	
terms found in Romexis "Free text notes"	43
Figure 8. Examples of different locations of root resorption	44
Figure 9. Example of Class 2 cases with well-defined borders and without a channeling	
spread	45
Figure 10. An excluded case, showing an example of false positive diagnosis of external	
cervical root resorption	46
Figure 11. Chart illustrates the 5 steps followed for the sample selection of the external	
cervical root resorption group	47
Figure 12. Examples of two untreated cases with external cervical root resorption, and	
the changes noted in size over the years	52
Figure 13. Distribution of the age of patients at the time of external cervical root	
resorption diagnosis	55
	xi

Figure 14. Bar chart illustrates the number of patients presenting different potential local
predisposing factors in the control and the ECRR groups
Figure 15. Pie charts illustrate the frequency of simultaneous presence of different potential
local predisposing factors in the control and the ECRR groups
Figure 16. Chart illustrating the number of patients who were or were not recently
examined at a follow-up appointment60
Figure 17. Chart illustrating the number of patients who were or were not examined
at a follow-up appointment
Figure 18. Distribution of teeth with external cervical root resorption
Figure 19. Distribution of teeth based on the location of the portal of entry or, when it is
not visible, the center of the resorptive defect
Figure 20. Distribution of teeth based on the Heithersay classification
Figure 21. Chart illustrating the number of teeth which did or did not receive initial
treatment of the external cervical root resorption
Figure 22. Clinical pictures of a tooth failed due to a periodontal problem
Figure 23. Treatment outcome and probability of failure during a 10-year follow up period70
Figure 24. Treatment outcome. Determinant: external cervical root resorption repair
Figure 25. Treatment outcome. Determinant: Type of repair of external cervical root
resorption repair
Figure 26. Treatment outcome. Determinant: Location of the tooth, anterior or posterior73
Figure 27. Treatment outcome. Determinant: Root canal treatment
Figure 28. Treatment outcome. Determinant: External cervical root resorption repair
combined with or without root canal treatment75
xii

Figure 29. Treatment outcome. Determinant: Location of portal of entry (POE),	
or if not visible, center of resorptive defect	.76
Figure 30. Treatment outcome. Determinant: Heithersay classification	77

List of Abbreviations

AAE	American Association of Endodontists
СЕЈ	Cementoenamel junction
CNS	Central nervous system
ECRR	External cervical root resorption
EPT	Electrical pulp test
FeHV-1	Feline herpes virus type 1
GE	Graduate Endodontics
GI	Glass ionomer
MTA	Mineral trioxide aggregate
OPG/RANKL/RANK	Osteoprotegerin/ receptor activator of nuclear factor kappa B
	ligand/ receptor activator of nuclear factor kappa B
PDL	Periodontal ligament
POE	Portal of entry
PRRS	Pericanalar resorption resistant sheet
RR	Root resorption
SLOB	Same lingual opposite buccal
TCA	Trichloroacetic acid
UBC	University of British Columbia
VRF	Vertical root fracture

Acknowledgements

I owe my deep gratitude to my supervisors Dr. Haapasalo and Dr. Aleksejuniene, who took keen interest on this project and guided me all along. I would like to heartily thank them for their encouragement, their timely support, and all the hours we have spent working on this study.

It is my honor to work with Dr. Haapasalo. His expertise and interest in the topic inspired me to engage with this project. I will always admire him for being humble and accessible to his students, while he is a worldwide known endodontist. I want to thank him for bringing an equilibrium of hard work and fun/excitement in the learning process.

The final outcome of this project required a lot of Dr. Aleksejuniene's expertise and experience, and I was fortunate to have worked with her. I also want to thank her for being an inspiring teacher and making this process challenging for me.

I am thankful for having Dr. Shen and Dr. Hieawy as my committee members. I would like to thank Dr. Shen for her gentle approach and for always having her door open for my questions. I would like to thank Dr. Hieawy for being such a motivating role model, for sharing his knowledge and for supporting me from my first steps at UBC, till today.

I heartily respect and thank Dr. Coil for being such an open-minded mentor for me. Every day I keep learning something from Dr. Coil. He has a unique way to motivate us, his students, to become our best selves. I am fortunate to have met and worked with Dr. Coil.

I am grateful for my "UBC family" at Division of Endodontics: my clinical instructors, my classmates, and the clinical staff who have made for me this experience pleasant and full of knowledge.

Dedication

This dedication is for my family.

To my mom and dad, for supporting my decisions with all their heart, for teaching me not to give up and for always contributing to my happiness with all their strength. You are the most inspiring role models I could have ever had. To my sister, Vasiliki, for being my best friend since I remember myself and for always standing by me. Your enduring support makes me overcome any obstacles. To my brother, Tasso, for always supporting me and believing in me. Your unwavering support has changed my life. Thank you for everything. To my love, George, for supporting me throughout my education and for being the best company in my life. Without you, this journey would have never been the same. Thank you for making this possible for us.

Chapter 1: Literature review

1.1 Introduction

Root resorption in primary teeth is a physiological process (1). However, in permanent teeth it is a pathological process which may result in loss of dentin, cementum and/or bone and may vary in appearance (2). More specifically, the different types of root resorption which have different clinical and radiographic appearance, and different etiology, are the following: internal inflammatory resorption, external inflammatory resorption, replacement resorption, cervical resorption and transient apical breakdown (2).

The present study is focusing on the external cervical root resorption (ECRR), which is typically located at the cervical area of the root and it is mainly invasive in nature (3). The ECRR starts as a small defect, but it may expand apico-coronally and/or circumferentially around the root canal space (4). As a result, the teeth to maintain their vitality and the patients are often symptom free until the later stages of the resorption process (5).

The exact etiology behind the ECRR is unknown. Two cohort studies have been published presenting ECRR potential local predisposing factors (3,6). However, there is no cohort study examining any systemic conditions as predisposing determinants. In addition, there is only one published study examining the treatment outcome of the ECRR cases (7). This study (7) examines the treatment outcome based only on one local factor (i.e. the Heithersay classification).

The current project will assess if any systemic determinants are associated with an increased prevalence of ECRR. In addition, it will assess if any treatment-related determinants are associated with lower rates of ECRR treatment failure, and if any local determinants are associated with a higher ECRR failure rates.

Understanding the root resorption

1.2 Classification of root resorption

Different attempts to classify root resorption (RR) have been recorded in an effort to organize the different types of RR into distinct categories. An indirect goal of this attempt was to identify the most effective modality of treatment in each category. The following is a brief historical review of different approaches regarding the classification of RR.

Historically, tooth resorption was first mentioned in the 16th century, in a text called "*Artzney Bunchlein*" (8). Since then, multiple different names have been suggested in order to describe external resorptive defects, such as "*spina ventosa*" (9), "*absorption of permanent fangs*" (10,11), "*rarefying periodontitis*" (12), "*idiopathic resorption*" (13), "*spontaneous intermittent resorption*" (14), "*burrowing resorption*" (15).

In 1970, Andreasen (16), approached the topic and classified the RR of injured teeth as follows: internal resorption, which could be inflammatory or replacement, and external resorption, which could be surface, inflammatory or replacement resorption. Two years later, Tronstad (17), classified the inflammatory RR into internal and external, and stated that it can have either transient or progressive character. Furthermore, the cervical and the replacement resorption were added to into his classification in addition to inflammatory. In 1999, Gunraj (18) described the types of RR according to etiologic factors. He stated that surface, inflammatory and replacement resorptions may be present after traumatic incidents and that external RR can be a result of either pulp necrosis and periradicular pathosis, or of pressure in the periodontal ligament (PDL). Internal and cervical resorption were described as well (18). The same year, Ne et al. (19) classified RR according to clinical and histological findings. Internal RR was subdivided into inflammatory and metaplastic (root replacement resorption). External resorption

was classified as surface RR, inflammatory (cervical or apical) RR, ankylosis, replacement RR (separately from ankylosis), and transient apical breakdown. Combined internal and external resorption, as separate individual classification of RR, was also proposed in this paper (19). Fuss et al., in 2003, proposed a classification according to factors that can act as stimuli. Their classification was as follows: pulpal infection RR, periodontal infection RR, orthodontic pressure RR, impacted tooth or tumor pressure RR, ankylotic RR (20). Another classification, which was introduced in 2006 by Lindskog et al. (21), was supported again one year later by Heithersay in a paper about the management of tooth resorption (22). This classification included three main categories: trauma induced, infection induced and hyperplastic resorption. Each category had also subcategories as follows: Trauma induced resorption could be: 1) surface, 2) transient apical internal, 3) pressure, 4) orthodontic, or 5) replacement.

Infection induced resorption could be: 1) internal inflammatory, 2) external inflammatory, or 3) communicating internal external.

Hyperplastic resorption could be: 1) internal (invasive) replacement or 2) invasive coronal (22).

The most resent classification which will be used throughout this thesis, is the American Association of Endodontists (AAE) classification in 2016 (2). More specifically, the resorption is defined as "*a condition associated with either a physiologic or a pathologic process resulting in a loss of dentin, cementum and/or bone*". The classification and how each term is defined, are as follows in Figure 1 (2):

American Association of Endodontists - Glossary of Terms

"Internal resorption — An inflammatory process initiated within the pulp space with loss of dentin and possible invasion of the cementum" (a)

"External resorption — Resorption initiated in the periodontium and initially affecting the external surfaces of a tooth; may be further classified as surface, inflammatory or replacement, or by location as cervical, lateral or apical; may or may not invade the dental pulp space"

"Surface resorption — A physiologic process causing small superficial defects in the cementum and underlying dentin that undergo repair by deposition of new cementum"

"Inflammatory resorption — An internal (a) or external (b) pathologic loss of tooth structure and possibly bone, resulting in a defect; occurs as the result of microbial infection; characterized radiographically by radiolucent areas along the root"

"Replacement resorption (ankylosis) — A pathologic loss of cementum, dentin and periodontal ligament with subsequent replacement of such structures by bone, resulting in fusion of bone and tooth" (c)

"Cervical resorption — A type of external resorption that occurs in the coronal third of the root" (d)

"Transient apical breakdown — A response to tooth luxation consisting of radiographic apical bone and root resorption that resolves spontaneously without intervening treatment"



Figure 1. Terminology and classification of root resorption according to the American Association of Endodontists (2). Illustration of different types of root resorption: a) internal inflammatory resorption, b) external inflammatory resorption, c) replacement resorption, d) external cervical resorption.

1.3 Histology of root resorption

Odontoclasts, which are associated with root resorption, exhibit important morphological and functional similarities with osteoclasts (23). These similarities have led some researchers to conclude that odontoclasts and osteoclasts are probably identical. The list of their similarities includes the same contiguous enzymatic function and cytologic characteristics. In addition, they both act on mineralized tissue by causing resorptive defects, known as Howship lacunae (24,25). Furthermore, they are both multinuclear cells, which are the result of mononuclear cell fusion (5,26). Even though mononuclear osteoclasts and odontoclasts have been found in bone and tooth resorptive defects respectively, it is well accepted that as the resorption propagates, multinuclear cells predominate (27,28).

Characteristics that have been suggested as distinguishing factors among these cell types; are the greater size of the cell and its sealing zone and the greater number of nuclei that can be identified in osteoclasts (29). Furseth (30) observed that under an electron microscope, odontoclasts' cytoplasm shows a more vivid staining result than that of its neighbouring cells. This is probably due to the great number of mitochondria, vacuoles and cytoplasmic ribosomes. Their nucleoli are big and symmetrically positioned in the middle of the cell. According to the same researcher, "where the odontoclast was in contact with the tooth surface, a system of canals extending 2-3 μ m into the cytoplasm was observed, and these canals contained mineral crystals" (30).

There are different approaches regarding the differentiation of resorptive cells precursors. According to Speziani et al. (31), immature dendritic cells can differentiate their operation into osteoclastic operation and since these cells are present in the pulp tissue, it can be implied that they can also act as odontoclast precursors. According to other researchers, the

5

OPG/RANKL/RANK transcription factor system has been implied to be responsible for the differentiation of the odontoclasts from their precursors. (32).

It is interesting to examine though, not only the differentiation of resorptive cells which is the molecular starting point for this process, but also the anti-resorption factors. According to Wedenberg et al. (25,33), the expansion of the internal resorption can be prevented by the presence of an odontoblast layer and predentin. Similarly, the expansion of the external resorption can be prevented by the presence of precementum (25,33). In addition, the resorption can only be active when the pulp tissue is vital and at the same time there is a constant bacterial proliferation, probably by partially necrotic pulp tissue (26).

The absence of stimulating factors is critical in order for the resorptive process to cease its progression. The stimulating factors for resorption, can present a concurrent action with the causal factors of the damage on the predentin or precementum of the root. This is a topic that has been examined in the past and it has been shown that there are many different factors (i.e. orthodontic treatment and history of trauma), that can lead to different kinds of resorption (i.e. external or internal) on different areas of the root (cervical, middle or apical third) (20), but it is out of the scope of this thesis to expand further on the topic of causal factors.

1.3.1 Physiological root resorption

Even if the stimulating signals for the physiological root resorption of the deciduous teeth are not well established, it is more likely that this process is a result related to the action of cytokines and transcription factors (1). More specifically, the applied force on the primary teeth by the underlying permanent teeth, is the stimulating factor that activates a sequence of

6

molecular responses on root and bone tissue, which results in the resorption. However, at the same time, the tooth structure of the permanent adjacent teeth is well protected and instead of being resorbed it develops normally. It is interesting to mention that there is no presence of bacteria during physiological root resorption (26).

1.3.2 Pathological root resorption

Unlike in deciduous teeth, resorption in permanent teeth is a pathologic condition. Permanent teeth cannot undergo physiologic remodeling like bone and regardless the fact that bone resorption may take place around them, they are not usually attacked by clastic cells. However, when this pathological root resorption takes place, the appearance may vary.

Internal root resorption

Throughout the decades, histological studies have been done on extracted teeth diagnosed with internal inflammatory RR, which was initially caused either by natural or experimental forces (34,35). Wedenberg and Zetterqvist (34), in 1987, examined both primary and permanent extracted human teeth and assessed the histologic features of the naturally propagated internal inflammatory RR. According to their findings, the only difference between the two groups of teeth, was that the resorptive defect was propagating faster for the group of primary teeth. Regarding the histological components of the pulp tissue, the researchers found lymphocytes and macrophages to predominate, and some neutrophils to be present. The normal pulp tissue had greater vascularity than the resorptive tissue which was similar to the fibrotic connective tissue of the periodontium. The dentinal walls of resorptive defects were predominated by multinuclear odontoclasts of great size. The adjacent connective tissue was predominated by mononuclear

cells, probably odontoclast precursors. It is interesting to mention that both multinuclear and mononuclear cells, presented tartarate-resistant acid phosphatase (TRAP) activity. It is also fascinating that on the root canal wall of all the examined teeth, mineralized tissue (bone or cementum-like) was present. Furthermore, mineralized tissue islands were present inside the root canal system, which was proposed to be a finding of odontoblast action (34).

External surface root resorption

This type of resorption differs from the others because there is no incessant stimulus. Even though it has been shown that the constant stimulation of the resorbing cells is a requirement for phagocytosis during the resorptive process (36), this kind of resorption is an exception. As a result of the absence of a stimulation force, after a period of a couple of weeks of resorptive action, this phenomenon stops and the area of concern will be repaired by a cementum-like tissue (17). These resorption defects are present without any PDL inflammation, on normal cementum, and may or may not be identified on a radiograph (37).

External inflammatory root resorption

The histologic appearance of the resorptive defects is bowl-shaped areas, closely related to PDL inflammation. This is probably a result of bacterial antigens from the pulp tissue through tubules. This inflammation results in a growing granulation tissue matrix, which is enclosed in the resorptive defect but at the same time is in contact with the adjacent PDL through the hard tissue loss. This matrix has been shown histologically to consist of plasma cells, lymphocytes and polymorphonuclear leukocytes. The inflammatory reaction can also expand itself into the adjacent capillaries. It is noteworthy that in histologic findings, the odontoclasts are located in the lacunae, in both cementum and dentin, but not in the granulation matrix (37).

Replacement resorption

This type of resorption was named based on the process in which the tooth structure has been replaced in a continuous way by bone. The initial formation of this relationship between bone and tooth structure has not been well described. Yet, on the occasion of a substantiate relationship, the tooth structure either is considered as a "*dead bone transplant*", so it is replaced, or it is considered as a "*part of the physiological bone remodeling*" (37).

External cervical root resorption

This type of resorption follows the pattern of the other resorptions, with the exception that it is typically located in the cervical area and it is mainly invasive in nature. The location is associated with the level of the junctional epithelium and the probing depths (38). At the beginning of the acute resorptive process, odontoclasts can be found in the lacunae and granulation tissue without inflammatory cells inhabiting the defect (3). However, bacterial proliferation is present at histological images of subsequent stages (4,39). An example of the ECRR histological findings is illustrated in Figure 2.



Figure 2. Histological findings. a) periapical radiograph of tooth #31 with external cervical root resorption (ECRR),b) pre-surgical intraoral photo (black arrow shows the tooth with ECRR), c) peri-surgical intraoral photo (yellow circle shows the resorptive defect), d) histological findings, e) magnified area of picture d.

The pattern of expansion for the resorptive defect through the dentin, can be either circumferential around the root canal system or apico-coronal (3). This defect expands itself without invading the root canal system during the first stages, since the presence of predentin acts as an anti-resorption factor as previously mentioned (25,33,40). However, communication between the resorptive defect and the root canal system or the PDL can be noticed at later stages. As a result, the adjacent pulp tissue or the periapical tissue respectively, has normal histologic appearance as long the resorption creates channels only though dentin (3).

The response to the resorptive propagation in some types of cervical resorption and at later stages, is the deposition of bone like tissue in contact with dentin. It was stated that this histologic finding shows an effort of reducing the loss structure (3).

1.4 Classification of external cervical root resorption

Historically, different terminology has been used to describe ECRR, such as "odontoclastoma" (41), "idiopathic external resorption" (42), "peripheral cervical resorption" (43), "progressive intradental resorption" (44), "cervical external resorption" (45), "late external resorption" (46), "extracanal invasive resorption" (47), "supraosseous extracanal invasive resorption" (48), "cervical resorption" (17), or "peripheral inflammatory root resorption" (38).

To analyze in depth the phenomenon of ECRR, it is important to understand the different patterns that can be identified in a clinical situation. In 1987, Fank and Bakland classified the *"extracanal invasive resorption"* as intraosseous or supraosseous, based on the possible location of the POE (48). A year later, Frank and Torabinejad further categorized the *"extracanal invasive*

rerorption", again based on the location of the POE, but this time they suggested 3 classes, i.e. intraosseous, crestal and supraosseous (49).

In 1999, Heithersay published a clinical classification of ECRR, which became the gold standard in the following years. This classification was based on the extension and the pattern of the resorptive defect, as grouped in Figure 3 (3):



Figure 3. Illustration of the Heithersay classification of external cervical root resorption (3).

In 2018, after the recommended use of CBCT as a means to determine prognosis and

treatment plan in root resorption cases (50), two different "three dimentional" classifications

have been proposed (51,52). The first one (51) recommended a descriptive classification based

on the combination of three measurements on the CBCT image: the height of the lesion, the

circumferential spread and the proximity to the root canal. More specifically, the authors proposed the use of a number from 1 to 4 in order to describe the height of the lesion (i.e. 1: at CEJ or supracrestally, 2: extension into coronal third and subrestally, 3: extension into middle third, 4: extension into apical third of the root). In addition to that, they proposed the use of a capital letter from A to D in order to describe the circumferential spread (i.e. $A \le 90^\circ$, $B: > 90^\circ$ to $\le 180^\circ$, $C: > 180^\circ$ to $\le 270^\circ$, $D: > 270^\circ$), and finally a lower-case letter was used to describe the proximity to the root canal (i.e. d: lesion restricted to dentine, p: possible pulpal involvement) (Fig. 4). According to the authors, the combination of the number, the capital and the lower-case letter will provide an effective way of sharing information between colleagues accurately. It seems to be a precise, albeit complicated way of classification, and it requires that a CBCT of good quality is available. The second paper (52) proposed the Rohde Classification System, which is based on the findings of the axial slides of the CBCT images. More specifically, this classification is focused on the amount of dentin loss in the cervical third and the external surface of the tooth. The proposed three classes are based on the circumferential extent of the resorptive defect:

class 1: in less than 1/3 of the tooth,

class 2: in less than 1/3 of the tooth with a perforation defect \geq 2.5mm in any dimension, class 3: in more than 1/3 of the tooth.

According to the authors, this classification system may be useful for treatment planning.

a b	c	1
Height	At the CEJ or supracrestal	1
Circumferential spread	≤ 90°	A
Proximity to the root canal	Lesion confined to dentine	d
Classification		1Ad

Figure 4. Illustration of a novel 3D classification (51). The yellow arrows point the external cervical root resorption.

a) periapical radiograph, b) sagittal CBCT view, c) axial CBCT view of a Class 1Ad case.

Throughout this thesis, the Heithersay classification will be followed.

Understanding the external cervical root resorption

1.5 Mechanism of action of external cervical root resorption

Two recently published studies provide information regarding the mechanism of action of ECRR in vital (53) and previously endodontically treated teeth (54). In both cases, the ECRR is progressing in three main sequential phases. The first phase is named the "*resorption initiation*", the second the "*resorption progression*", and the last one the "*repair*" phase. It seems that resorption and repair can take place simultaneously on a root, but at different areas of it. Even if the phases are common in both vital and previously treated teeth, considerable differences between them exist.

1.5.1 External cervical root resorption in vital teeth

Initially, localized PDL breakdown takes place (38) leading to inflammation limited to that area. Activated macrophages travel to the breakdown area and assist in the formation of granulation tissue (55). A potential uncovered dentin area at the level of the cementoenamel junction (CEJ) could be vulnerable if it comes into contact with the granulation tissue (56). If indeed the dentin is exposed at the level of the CEJ, then this area is a potential portal of entry (POE) for cervical resorption. There are three possible scenarios that can take place, based on the type of tissue adjacent to the POE (57). The first scenario takes place when the injured area is reconstructed by bone cells, which results in replacement resorption (55). In the second scenario, the area is reconstructed by PDL cells that results in the regeneration of the periodontium (55). The last scenario, in which there is no repair, takes place when gingival connective tissue comes into contact with the exposed dentin (55). According to Mavridou et al. (53), PDL is usually not present at the POE of ECRR, but the area is predominated by fusion and augmentation of the alveolar bone into the resorptive lacunae. However, a stimulus, such as infection or mechanical force, is required in order for the ECRR to propagate (38). It has been hypothesized, that hypoxia at the resorptive microenvironment possibly plays a significant role at this first stage of ECRR (53).

During the second phase, the resorption spreads towards the pulp, by tearing down cementum, dentin and enamel; without invading the pulp space per se, due to a resistant layer. It then creates resorptive channels in all three dimensions and expands with a more circumferential spread. By using micro-CT, the mean thickness of the residual circumpulpal/pericanalar dentin has been found to be 210 μ m, and the contrast of this area has been shown to differ from the rest of the dentin; which could be interpreted as an area with a distinctive mineral substance (58). According to Mavridou et al. (53), the pulp plays a role during the resorptive process by propagating calcification of the extracellular matrix, especially at the areas close to the active sites of ECRR. The authors named the residual circumpulpal layer as pericanalar resorption resistant sheet (PRRS). They advocated that the occurrence of the PRRS of this high impedance is due to two main reasons. Either due to vitality of the pulp, which provides normoxia in the PRRS, versus hypoxia in the resorptive areas, or due to the way the mineralization of this layer is distributed, being more mineralized at the external surface and gradually less mineralized in the internal surface (53).

During the third stage, bone like tissue travells through the POE and grows into the resorptive lacunae towards the pulp, a process that has been interpreted as healing (14,53,59). Also, in areas where the PDL is disintegrated, a localized fusion between the neighboring alveolar bone and the newly formed bone like tissue takes place. It has been noticed that osteoid tissue apposition and remodeling processes can coexist (53).

16

1.5.2 External cervical root resorption in endodontically treated teeth

During the initiation stage of ECRR in endodontically treated teeth, all the steps described for the vital teeth take place. The main difference is that in previously treated teeth, there is no bone tissue coming through the POE into the resorptive lacunae, and as such there is no fusion between the root and the adjacent alveolar bone (54). An explanation supported by the authors, is that due to lack of vitality these teeth are also lacking the balanced action of normoxia; and as a result hypoxia predominates (54). It has also been suggested that under hypoxia, angiogenesis is high which potentially leads to a successive growth of a highly vascularized granulation tissue in the area (60).

During the second stage, a similar pattern of ECRR has been noticed in both vital and previously treated teeth. The main difference is that the resorption proceeds in a more expansive way for the endodontically treated teeth. Two potential explanations have been given, the absence of a PRRS (53,54) and the different chemical composition of root dentin after its exposure to irrigants during the RCT (54,61).

During the third stage, remodelling still takes place, similarly to the vital teeth, but it is less extensive (54).

1.6 Etiology and predisposing factors

The precise etiology of ECRR remains a mystery that has not been solved. However, it is generally recognized that a deficiency or damage on the cementum layer at the cervical area of the root, bellow the epithelial attachment, could be described as a requirement for a resorption to initiate (38). This has been confirmed by animal studies (55). This deficiency on cementum layer can be either a result of developmental insufficiency that allows a POE at the CEJ level (62), or a result of chemical or mechanical breakdown (54,56).

17

In an effort to identify the predisposing factors of ECRR, several studies have been conducted. However, due to the complex character of the topic, the vast majority of these studies were case reports and case series. Hitherto, only two cohort studies have been carried out (3,6). In this section all these studies will be analyzed in a comprehensive way, with an emphasis on the cohort ones.

To the best of our knowledge, there are no published cohort studies examining systemic determinants as potential predisposing factors for ECRR. However, in this section case reports suggesting an association between systemic conditions and ECRR will be presented. Two systemic conditions (i.e. the use of bisphosphonates and viral infections) were examined as predisposing factors for ECRR in a recent cohort study (6). However, this study included these factors based on existing literature and more specifically based on published case reports. That emphasizes the lack of cohort studies examining which systemic conditions could potentially be predisposing factors for ECRR.

1.6.1 Predisposing factors based on cohort studies

Heithersay published a retrospective study in 1999 (3), where 257 teeth diagnosed with ECRR in 222 patients were analyzed for the presence of possible predisposing factors to ECRR. More specifically, information regarding trauma, intracoronal bleaching, surgery in the area, orthodontic treatment, periodontal root scaling or planing, bruxism, delayed eruption, developmental defects, other treatments or incidents, and intracoronal restorations was collected. Also age, gender, medical and dental history were recorded as well. According to the results of this study, orthodontic treatment was the most common predisposing factor (24.1% of the teeth). Trauma and intracoronal restoration were the next on the list (15.1% and 14.4% of the teeth, respectively). Surgical procedures at the level of the CEJ and intracoronal bleaching were other two factors that have been associated with the occurrence of ECRR (5.4% and 3.9% of the teeth, respectively). All the other factors that were analyzed in this study were present with low frequency and were the following: bruxism (2.4% of the teeth), eruption disorders (1.6% of the teeth), developmental disorders i.e. grooves or invaginations (0.8% of the teeth), and interproximal stripping (0.6% of the teeth). However, another important finding of this study is that in 16.4% of the teeth with ECRR there was no correlation with any predisposing factor. The author commented that those teeth "*may have had undetectable developmental defects of cementum*", and therefore categorized the ECRR as idiopathic. In 28.9% of the cases more than one predisposing factor was present, a finding that suggests a multifactorial character of the ECRR (3).

Mavridou et al. (6), collected data from 337 teeth diagnosed with ECRR, in 284 patients. Overall, the evidence was tested on the following potential predisposing factors: cracks, poor oral health, developmental disorders, malocclusion, frenulum tension at cervical gingiva, eruption disorders, extraction of adjacent tooth, nonvital bleaching, orthodontic therapy, orthognathic surgery, parafunctional habits, periodontal surgery, previous traumatic injury, restorative and endodontic procedures, systemic diseases, medication, viral infections and playing of wind instruments. The authors included items to the list based on the existing literature. Orthodontic therapy was the most common factor (45.7% of the teeth). Trauma, parafunctional habits and poor oral health were frequent factors as well (28.5%, 23.2% and 22.9% of the teeth, respectively). Less frequent factors were malocclusion (17.5% of the teeth) and extraction of an adjacent tooth (14.0% of the teeth). Other factors with a low rate of frequency but still noteworthy were the following: eruption disorders (2.7% of the teeth), cracks

19

(2.1% of the teeth), developmental disorders (1.5% of the teeth), intracoronal restorations (1.2% of the teeth) and frenulum tension at the cervical area (0.6% of the teeth). An interesting observation was also that 59.0% of the cases had more than one predisposing factor, potentially indicating a multifactorial nature for ECRR. Gender appeared to play no role as a predisposing factor for ECRR, while the tooth site seemed to have an impact, with maxillary central incisors being the most frequent diagnosed teeth. Age showed to have a different distribution for each potential predisposing factor. For instance, when orthodontics was the examined determinant, the majority of the patients were found in the age group of 15-19 years, whereas when the determinant was poor oral health, the majority of the patients were found in the age group of above 65 years (6).

1.6.2 Predisposing factors based on case and case series reports

Orthodontic treatment

In 1984, a study on 87 post-orthodontic patients, concluded that ECRR is not a frequent complication since only one patient presented with it (63). This is in agreement with a more recent 8-year orthodontic follow-up study, where only one out of 108 patients was diagnosed with ECRR (64).

Two other studies (65,66) followed an experimental model (67) and applied a variation of forces for different time intervals on teeth that would be later extracted based on the orthodontic treatment plan. There was a control group and all extracted teeth eventually underwent volumetric analysis (65,66). The first study (65) was published in 2006, and assessed the physical properties of root cementum. This study shed light on the frequency of ECRR occurrence, based on the different surfaces of the root in association with different types of force

20
applied (i.e. compression, tension). The heavy-force group (225g of buccal tipping orthodontic forces) was found to have eight times the amount of resorption at the buccal cervical area, than the light-force group (25g). When the middle and apical surfaces of the root were examined for different force levels, there was no significant difference for the occurred RR (65). The second study in 2017 (66), showed that the extent of ECRR was related to the extent of the tooth movement. Significant ECRR was noted when, for a period of two months, a force of 2 N was applied. Moreover, the mandibular teeth showed more resorption than the maxillary ones (66).

A case series in 2013 examined the influence of failed orthodontic treatment on impacted canines (68). The sample consisted of 14 patients that were referred by other orthodontists after the failure to move impacted canines. The number of teeth examined was 15 since one patient had two canines with ECRR, bilaterally (68). The presence of more than one tooth with ECRR in orthodontic patients has been reported more recently as well (69). It is noteworthy that the ECRR on the impacted canines presented regardless the severity of tooth displacement, and was already class 3 or 4 when patients were referred for a second opinion (68). It is interesting to mention that most of the teeth underwent an open surgical approach during the orthodontic treatment, which means that either mechanical forces or orthophosphoric acid leakage could have potentially initiated a cementum breakdown at the CEJ area. All teeth in this study had to be extracted and the authors made a clear point by concluding that early diagnosis of ECRR is critical (68).

Trauma

ECRR is a frequent unfortunate result of traumatic injuries on teeth (3,6,70). Cvek (46) has described it as "*late*" external resorption because it can be identified during a clinical examination years after the traumatic event took place. It was suggested that patients with a history of dental trauma should continue to be followed up radiographically (70), even when they do not have clinical signs or symptoms. Patients with history of intrusion and avulsion have a higher likelihood of presenting root resorption (not specifically ECRR) after a traumatic incident (71,72), and the need of strict follow-up is higher. The multidisciplinary approach seems to be a requirement for a good treatment outcome in more complicated trauma cases (73).

Bleaching

An unfortunate aftermath of RCT is sometimes the crown discoloration (74), and a common solution offered to this problem is bleaching. However, bleaching may have another unwanted consequence, which is the occurrence of ECRR. One of the first and well described predisposing factors for ECRR is bleaching of pulpless teeth (75). Harrington and Natkin (75) in 1979, presented a case series study where patients had a history of trauma and the RCT was done at a young age. These patients underwent bleaching with a caustic agent and heat source 6 to 15 years (with one exception of 6 months) after trauma. Incident of a second trauma was excluded based on the dental history. The fact that both mesial and distal cervical areas were affected in most of these cases and the timewise close proximity of ECRR to the bleaching procedure supported the hypothesis of predisposition. Two factors may have been involved: first, Superoxol may have diffused through patent dentinal tubules to reach the cervical PDL. Secondly, in 10.0% of the teeth the dentin is exposed to PDL tissue because the enamel and cementum do not meet

(76). These two factors were discussed in "Inflammatory resorptive response in the cervical area appears a reasonable hypothesis" by the authors, in addition to the possibility that heat at the cervical area can also be a causative factor for ECRR (75). Four years later, another research team supported this hypothesis (77). The main two differences between the two studies is firstly that this case series included teeth that underwent RCT while the patients were adults, and secondly the absence of history of trauma in the patients' dental history (77). Similar cases of ECRR after bleaching have been reported (4,78-83). Superoxol is "a 35% hydrogen peroxide" formula to bleach non-vital teeth that can be mixed with sodium perborate for the walking bleach method" (84), as stated in the manufacturer's web page. It is not presently known what the role of sodium perborate could be in the initiation of the resorptive process. In 1998, an *in vitro* study suggested an answer to this question by assessing the adherence index of sodium perborate on rat inflammatory macrophages. Due to its inhibitory action on macrophage adhesion, sodium perborate may be involved in the process of ECRR (85). A recent case report in 2017, presented a patient diagnosed with multiple ECRR on mandibular and maxillary incisors, after using a home bleaching kit with 22.0% carbamide peroxide-based gel for 4 nights consecutively (86).

In 1988, Friedman et al. (79) found that occurrence of ECRR after bleaching, and without history of trauma in 6.9% of cases (4 out of 58 patients). They drew the attention to the appropriate bleaching case selection and most importantly to the significance of preventive measures and follow-up (79). As a response to the need for prevention, different approaches have been proposed for intracoronal bleaching, in order to avoid ECRR (77,82).

Traumatic occlusion

In a case report of ECRR on tooth #21, a patient presented with traumatic occlusion on that tooth. Tooth #21 was labially displaced and had occlusal interference throughout protrusive movements and fremitus on centric occlusion. However, this specific case also presented with poor oral hygiene and a pyogenic granuloma in the area of the ECRR. (87). Occlusal trauma with the presence of periodontal findings (i.e. localized gingival overgrowth) had been reported previously as well (88). Malocclusion has frequently (17.5%) been found to be a predisposing factor for ECRR (6).

Periodontal findings

Poor oral hygiene or the presence of plaque is a factor often mentioned in cases of ECRR (87–91) and is well described as a significant possible predisposing factor (22.9%) for ECRR (6). Periodontal therapy (i.e. deep scaling including the area of the CEJ) has been seldom mentioned as a predisposing factor (3). Surgical periodontal therapy, on the other hand, has been found to be present as a predisposing factor from 1.8% of the teeth (6) to 5.4% of the teeth (3). Other periodontal findings, such as pyogenic granulomas (87), peripheral giant cell granulomas (recurrent or not) (38,92), localized gingival overgrowth (88) and gingival inflammation with crestal bone resorption (93) have also been associated with ECRR.

Extraction of a neighboring tooth

Extraction of a neighboring tooth has been described as a predisposing factor, since mechanical trauma to the remaining teeth can be caused during such procedure. A case report was published where a tooth adjacent to an extracted mandibular molar developed ECRR (94). However, in this particular case, orthodontic movement and a final tilted position of the tooth

(i.e. increased possibility of occlusal trauma), were reported after the extraction of the adjacent tooth. In a prospective cohort study, extraction of a neighboring tooth was found to have 14.0% frequency among the predisposing factors (6).

Playing wind instruments

A case series study hypothesized that pressure forces applied on the cervical area of the root by a wind instrument could trigger the initiation of ECRR (95). This hypothesis was supported by a cohort study which found 9 out of 337 teeth (2.7%) were diagnosed with ECRR in patients who play wind instruments (6). Another case of a clarinet player having ECRR on a lateral incisor was also published in 1992 (38).

Orofacial cleft

A retrospective cohort study shed light on the prevalence of ECRR in orofacial cleft patients. Out of 558 patients examined, 14 were diagnosed with ECRR (2.4%), six with unilateral and eight with bilateral cleft lip and palate. All 17 teeth affected were maxillary anterior teeth. All patients had undergone orthodontic treatment, and a significant number of them presented with dental history of periodontal surgery (n=12) or osteotomy (n=7) (96).

Bisphosphonates

In an effort to identify etiologic factors, Patel and Saberi (97) examined the effect of specific systemic medications. As they stated in their case series paper, bisphosphonates were related with "*an acute phase response and the release of proinflammatory cytokines*", which could be identified as a possible causal factor for ECRR.

Viral infections

There are three different types of viral infections that have been related to ECRR, the Varicella zoster virus, the Feline herpes virus type 1 (FeHV-1) and the Hepatitis B virus infection.

The varicella zoster virus was first associated with internal root resorption in 1986 (98), and then later in 2007 (99) and 2015 (100). It was only in 2016, that the varicella zoster virus was associated with an ECRR case (101). In this case report (101), a patient was hospitalized due to a severe trigeminal (right V3 branch) herpes zoster episode 19 years prior, and a mild anesthesia in the right lower area of patient's face was reported. The day that the patient was examined (19 years after the hospitalization), all teeth from the central incisor (#41) to the second molar (#47) located on the forth quadrant (right side of dentition of lower jaw) did not respond to vitality tests (both EPT and cold test). With one exception of a previously treated tooth, all other teeth were found to be without any clinical indication of necrosis, such as deep caries. As a consequence of the zoster virus, multiple periapical radiolucencies presented in the affected area, along with a ECRR in the mandibular canine of the affected side. It is noteworthy that the existence of varicella zoster virus in the RC system, with the absence of bacteria, was confirmed by molecular and culture tests. Other predisposing factors, such as history of trauma or orthodontic treatment were excluded according to the patient's dental history (101). Based on a review paper (102), there are no other reported cases of varicella zoster virus detected in patients with any kind of root resorption.

The second type of viral infections related to ECRR is the Feline herpes virus type 1 (FeHV-1). An interesting observation is that feline (domestic, captive and wild ones) odontoclastic resorptive lesions in the cervical area of teeth is a frequent clinical finding for

veterinary dentists (103,104). In a case series report (105), four patients were diagnosed with multiple ECRR and no predisposing factor was found on their dental or medical history. The authors having the background knowledge of feline odontoclastic resorptive lesions and in an effort to identify the etiology of ECRR, asked their patients whether they had been an encounter with cats. They received a positive response from all the patients, and blood samples were collected for neutralization testing of FeHV-1, which verified the transmission. This report highlights the possibility of transmission of FeHV-1 to humans (105).

The third and most recently published type of viral infections related to ECRR is the Hepatitis B virus. This case reported on a patient where a multiple idiopathic ECRR affected 20 teeth. No other predisposing factors, except the viral infection, were found in the medical or dental history of the patient (106).

Systemic sclerosis

A case report of a patient with multiple ECRR lesions identified a potential correlation of ECRR with systemic sclerosis, after other possible predisposing factors were excluded based on the patient's dental history (107). Systemic sclerosis is an autoimmune disease, characterized by microvascular breakdown and generalized fibrosis. The authors described a potential correlation between the high osteoclastogenesis (108,109) of the mandible and the distal phalanges of fingers (a result of the systemic sclerosis), with the mechanism of action of ECRR (107).

Familial pattern

The effect of genetic predisposition has been discussed in a case where a father and his son and daughter, two out of his six children, were all diagnosed with multiple idiopathic ECRR (110). In the study of Gold and Hasselgren (38) published in 1992, the authors made a comment

about a potential link between idiopathic ECRR and genetic factors. According to the authors, idiopathic cases of ECRR have been diagnosed in family members.

Diet

According to a case report, a patient developed ECRR lesions after having an acid enriched diet for two years (94). The patient also presented inflammation of the gingiva and resorption of the crestal bone. The authors proceeded with periodontal treatment and recommended alterations to the patient's diet. At the one year follow-up examination, repair of the resorptive defects with bone-like tissue was noticed (93).

1.6.3 Idiopathic external cervical root resorption

When none of the known predisposing factors can be identified through the patients' dental or medical history or by a thorough clinical examination, ECRR is diagnosed as idiopathic (13,111,112). It is notable that when Heithersay (3) investigated ECRR potential predisposing factors, 16.4% of the teeth (15.3% of the patients) examined were not related to any of the investigated factors. Case reports of idiopathic ECRR (113) have been previously published. However, a recent cohort study (6) which examined a wide range of predisposing factors as they were indicated by the literature, concluded that known predisposing factors were not identified in only 1.0% of the cases.

Multiple idiopathic external cervical root resorption

Multiple idiopathic resorption is an extended version of ECRR. It was, initially reported in 1930 (114). It can affect from 3 (115) or more teeth per patient (106,115,116), to all of patient's teeth in extreme cases, including even impacted third molars (117). No etiological

factors have been identified through these patients' medical and dental histories. However, a potential link to the feline virus (105) and the Hepatitis B virus (106) has been suggested as a potential factor. A hereditary link was also proposed (110). Multiple idiopathic ECRR is a rare condition, and only 25 case reports or case series reports, including 34 patients, have been published between 1930 to 2018 (93,106,118–120). Multiple idiopathic ECRR has been more frequently diagnosed in female patients (121). Multiple idiopathic root resorptions at the middle and apical third of the root have been reported as well (122,123).

1.6.4 Multifactorial etiology

The process of understanding the etiology of ECRR is fascinating not only because of the mystery behind it, but also due to the common coexistence of more than one etiologic factor simultaneously. As mentioned earlier, it has been found that the multifactorial character of ECRR is a frequent finding (28.9% of the patients (3) and 59.0% of the cases (6)). Patients presenting with two etiologic factors (38.0% of the cases) are nearly as common as those presenting with one factor (40.0%). Three factors were present in 17.0% of the cases and more than 3 factors were found in 4.0% of the cases (6).

1.7 Diagnosis

The clinical and radiographic findings of ECRR can present a substantial variation regarding the dimensions and the location of resorptive defect, which depends on the class/ stage of ECRR (5).

1.7.1 Clinical findings

Patients diagnosed with ECRR are often symptom free, i.e. ECRR is frequently only an incidental radiographic finding (5,38,47–49,90,124–129). This is supported by a cohort study that found the majority of patients with ECRR, diagnosed when the resorptive defect was already at class 3 (3). A vital pulp is a frequent finding in ECRR cases too (17,38,47–49,53,89,90,94,95,124,126,127,130–138) and a pink spot on a crown (Fig. 5) can be present (5,94,95,124,126,129,130,134,135) as well.





However, a clinician should not rush to diagnose ECRR based on a pink spot on the crown and the positive response to vitality tests, because that can also be a clinical finding of an internal inflammatory root resorption (48,139). The presence of discoloration of a crown can be related to the location of the resorptive defect (49). When the defect is located interproximally it is more difficult to identify it. The POE of the resorptive defect can be associated with the presence of granulation tissue, and therefore localized bleeding on probing frequently occurs in ECRR cases (38,49). However, the POE is not always easy to locate, even after elevating a surgical flap, because it is often very small (even $< 1 \text{ mm}^2$), and because it is not necessarily associated with periodontal bone breakdown and apical migration (49).

Clinically, it is important for a clinician to be able to diagnostically differentiate ECRR from root caries. The first sign, according to Singla (137), is the presence of bleeding on probing in ECRR cases due to the granulation tissue related to the area. The second clinical sign during probing, is that the bottom of the "concavity" is hard in ECRR and soft in caries, when examined with a sharp dental explorer. In addition, in cases of root caries the root surface often shows yellow discoloration around the area. The third clinical sign is that ECRR defects are rarely exposed clinically, while root caries is usually accompanied with recession of the gingival tissue (137).

1.7.2 Radiographic findings and tools

Periapical radiographs are crucial for the ECRR localization, diagnosis and differential diagnosis, albeit these provide only a two dimensional view. In order to overcome this limitation, a clinician can take a combination of two periapical radiographs, one straight and one with a horizontal angulation, and combine the findings from these two views. The multiple angulation periapical radiographs can be used to differentiate internal RR from ECRR, by following the continuation of the root canal, which is intact in the ECRR cases while it is expanded in internal RR cases (38,137). In ECRR, the outline of the root canal remains uninterrupted even if it is superimposed by the defect (48,49). The horizontal parallax radiographic technique, following the SLOB rule, can be used to check if the resorptive defect remained centered on the root canal

space on multiple (more than one) radiographs taken with different angulation, which indicates internal root resorption (26,137,140). In contrast, in ECRR, when different radiographs are taken with a more mesial or distal angulation, the defect will move accordingly and will be projected on the external surface of the root (49,137). Another radiographic finding in ECRR of endodontically treated teeth, is that the root filling material follows the normal shape of the root canal, without expanding into the resorptive defect (49). Also, ECRR defects tend to be asymmetrical, while the internal RR defects are more often symmetrical (137).

ECRR can be present as a mottled area on the radiograph due to depositing of osteoid tissue (15). ECRR can also look like a caries-like lesion in a radiograph(43), which is why differential diagnosis is critical. In addition to multiple radiographs with different angulation of one tooth, a full mouth radiographic examination has been suggested as a diagnostic tool in cases of multiple idiopathic resorption (5). Another advantage of taking periapical radiographs with more than one angulations is the ability to rule out the presence of cervical resorption or cervical burnout, and avoid possible overtreatment (75,141).

According to Andreasen et al. (142) *in vitro* study, periapical radiographs are reliable for the diagnosis of an external root resorption only when the size of the defect is bigger than 1.2 to 1.8 mm in diameter and 0.6 to 0.9 mm in depth. None of the "small" artificial cavities, with dimensions 0.6 mm x 0.3 mm, were detectable on the periapical radiographs. This study also proved that high contrast and multiple horizontal angulations provide a diagnostically superior result; when compared to lower contrast and single angulation radiographs. Another conclusion of this study was that resorptive defects located proximally (mesio-distal plane) were easier to detect than those on the bucco-lingual plane when periapical radiographs are used (142). More recent *in vivo* (140) and *ex vivo* studies (143–145) have demonstrated that periapical radiographs

are not always good enough in detecting and providing adequate information about natural or artificial external resorption cavities because of geometric distortions and two dimensional views.

Despite the fact that periapical radiographs provide an acceptable level of accuracy regarding the diagnosis of resorptive defects, CBCT is more trustworthy for an accurate and early diagnosis which can lead to better treatment planning and more predictable clinical results (50,146). A detailed dynamic view that provides information about the geometrical associations in all three dimensions simultaneously, allows a more thorough evaluation (147) of the defect. By "reading" carefully a CBCT scan, a clinician can gain information about the size of the ECRR, its circumferential spread and location, how close it is to the root canal system, and if the ECRR can be approached from outside or inside the tooth. By having this information, a clinician has all the necessary tools for optimal treatment planning. A clinical study which included 115 patients diagnosed with ECRR and compared the clinical value of their periapical radiographs and CBCT scans, concluded that a CBCT provides information for better treatment planning in ECRR cases (148). In addition, a CBCT scan visualizes, in a three-dimensional way, the resorptive channelling formation and the potential communication with the PDL. The channels can contain granulation tissue, resorptive cells, bone-like tissue and vasculature (148,149). In this context, it becomes clear how important it is for the clinician to be able to visualize in advance the extent of the resorptive defect, in order to prevent any active resorptive remnants hidden by the newly formed bone-like tissue in the main lesion and the tiny channels, which could lead to treatment failure (150). The diagnostic superiority of CBCT over periapical radiographs in ECRR cases has been documented by multiple clinical studies (51,148), and in some of them the treatment plan was changed after a CBCT was taken (52), or the likelihood of

correct treatment planning increased (23). Furthermore, in another clinical study, periapical lesions related to ECRR cases were identified only after the CBCT was taken (150). In agreement, *ex vivo* studies have shown that CBCT scans were more likely to localize (143) and correctly identify the type (classification) of ECRR defects (145), regardless of the tooth and dimensions of the defect (144). Multiple case reports and case series publications have used the CBCT as a means for better treatment planning (91,94,95,125,126,131,132,151–153).

1.8 Treatment management

The management of ECRR is fascinating and can be challenging as well, if one considers that each ECRR case is unique regarding significant factors such as the location, the accessibility, the extension of the resorptive defect, or the potential symptoms and the chief complaint of the patient. In 1987, Frank and Bakland (48) described in detail the suggested treatment in cases of supraosseous ECRR, where the POE is detected within the periodontium. Specifically, it was recommended that after confirming the diagnosis of ECRR as described at the diagnosis section earlier, to surgically expose the resorptive defect area and curate the resorptive granulation tissue by using slow-speed burs until only sound dentin is left on the defect's walls. The use of spoon excavators was recommended in order to avoid any accidental disturbance of the root canal continuity with the bur. After that, the use of hard-setting calcium hydroxide paste and silver amalgam was recommended as a means of restoration. Reposition of the flap, education of the patient how to avoid plaque formation, and finally follow-up appointments were suggested (48). Even if RCT was not strongly recommended during those steps, one year later, Frank and Torabinejad (49) made the comment that simultaneous RCT may be beneficial since it may be required later. This latter article also provided a well-organized

review of treatment options, beyond the supraosseous ECRR cases. Specifically, for the crestal ECRR cases where the POE is detected at the crestal level, a surgical approach after the RCT is recommended, because the internal approach may result in probing defects. For the intraosseous cases, where the POE is detected within the crestal bone and there is no contact with the oral cavity, the authors provide two options depending on the accessibility: an intracanal approach through the access cavity or intentional replantation. The intracanal approach, as described, has as its main goal to access the POE from the pulp chamber and/or coronal root canal, then completely clean the resorption area by chemomechanical curettage, and hermetically seal the POE. It is important not to gauge overzealously the root with the long shank round bur, in order to minimize damage to periodontal tissues and any structural weakening of the root. After all these steps have been followed, the resorptive defect can be filled with GI, super-EBA or MTA. The authors made the comment that unsuccessfully cases treated intraorally, end up in failure due to bacterial contamination. In ECRR cases when none of the previously described treatment methods is attainable and when the resorptive defect is not extensive making the extraction in one piece feasible, intentional replantation can be offered as a treatment option and can be discussed thoroughly with the patient (49). An updated paper published by Frank (154) in 1995 is in agreement with all the above steps.

Since 1999, when Heithersay (3) introduced the ECRR classification system, most treatment recommendations are based on the class of each ECRR case. In general, as it was recommended by Haapasalo et al. (155), the main objectives for an effective treatment are: "*i*) *remove the resorptive tissue, ii*) *prevent the return of the resorption; and iii*) maintain or regain a strong tooth structure".

1.8.1 External cervical root resorption defects Class 1 and 2

An external approach is widely recommended for these categories of ECRR cases. One way to proceed, is after raising a full thickness flap and removing the resorptive tissue by using a curette or a bur, to restore the defect with a material of choice and reposition the flap. In this way, the blood supply to the resorptive cells is eliminated (71). Another way to proceed, is to apply a protective stroma of glycerol on the surrounding soft tissue, and then apply 90.0% aqueous solution of TCA on small cotton pellet for 1-2 minutes with mild pressure. This stops bleeding and results in coagulation necrosis of the resorptive tissue, which can then be easily removed. After adjusting the margins of the resorptive defect, it is filled e.g. with GI and then a layer of light-activated bonded resin and composite filling (5).

1.8.2 External cervical root resorption defects Class 3 and 4

Class 3 and 4 ECRR lesions are larger and more spread than types 1 and 2. They also have narrow, vertical channels extending towards the crown and the root. The size and shape of class 3 and 4 lesions makes them clinically challenging to treat. The following treatment options should be considered: no treatment (follow up with annual checks), internal approach and external surgical approach. In cases where the internal approach is chosen, the clinician should try to reach the POE, induce coagulation necrosis by applying TCA, and fill the defect with an appropriate material e.g. GI or MTA (5). In cases where there is a large POE and destruction of adjacent bone, an external approach should be considered either as the sole treatment or together with RCT and an internal approach. In the external approach, a full mucoperiosteal flap is raised and the granulation and osteoid tissues are cleaned out through an access point at the most apical part of the defect, in order to preserve some cementum and increase the chances of PDL re-

attachment to the root. One of the main difficulties in type 4 ECRR treatments, is loss of tooth substance and the difficulty to detect where osteoid tissue ends and healthy dentin starts. After removing the resorptive soft tissue and the hard bone-like osteoid tissue from the defect, it can be filled with the material of choice (71) or can remain unfilled (156). The root surface is covered with bone grafting material and after the area is covered with a Goretex membrane, the flap is re-attached (71). In this way, Rankow introduced the advantages of guided tissue regeneration to ECRR cases that require endodontic surgery (156). Another option, as mentioned earlier, is the combination of external and internal approach, with elective RCT (5,71). A two-step external/internal approach has been described, where during the first step the resorptive tissue is repositioned. At a later time, the resorptive defect is removed from the root resorptive defect through an internal access restored with MTA, but without raising a flap. It has been advocated that this should be the technique of choice in order to preserve the periodontal attachment (71).

Complementary options can be forced eruption (orthodontic extrusion), in cases with long roots (5,20,71), or forced eruption combined with re-intrusion, when the crown-root ratio is not adequate (71), or even intentional replantation in cases with low risk for root fracture during extraction (71).

In summary, as recommended by Haapasalo et al. (155) in the most recent paper providing treatment guidance, for the class1 and 2 ECRR cases the external (surgical) approach is the treatment of choice. In these cases, resin composite or GI can be used to fill the resorptive defect. However, when the portal of entry is small and in proximity to the pulp space, the internal approach (through the root canal) is recommended, in combination with either bioceramic cement, GI or composite. For class 3 and 4 cases, which have guarded or poor prognosis, RCT

and an internal approach for the resorptive defect is recommended together with use of bioceramic cements.

1.8.3 Treatment considerations and prognosis

The American Association of Endodontists, in the Position of Statement regarding the *"Treatment options for the compromised tooth: a decision guide"*, have included a table which summarizes all the significant points a clinician should consider before providing any treatment option to their patient diagnosed with external resorption. For instance, in cases of minor missing tooth structure and with straight access to the supraosseous lesion, the prognosis is favorable. On the other hand, when the majority of the root structure is resorbed, not accessible surgically and associated with periodontal problems, the prognosis is unfavorable (157).

Even though many different approaches have been described regarding the management of ECRR cases, there is no specific gold standard, which a clinician can follow. A review conducted by the Cochrane Collaboration group regarding the management of external root resorption of all types, emphasized the lack of credible evidence. Consequently, it was suggested that dentists should use their clinical experience, based on patients' needs and considering each case individually (158).

1.9 Treatment outcome

To the best of our knowledge, there is only one published study that examined the treatment outcome of ECRR cases. This study was done in 1999 by Heithersay (159), and analyzed the clinical results of 94 patients, with 101 affected teeth, that have been treated non surgically. The first step was the protection of the surrounding soft tissue by application of glycerol, then a topical application of 90.0% aqueous TCA and the subsequent removal of the resorptive tissue. Following that, the defect was filled with GI. RCT was provided only when it was required. In this study, the patients had to have attended a follow-up of 3 years or more in order to be included. An exception was made for the cases that failed in less than 3 years, which were also included. The cases were divided in four groups, according to the Heithersay classification (3). Overall, 4 teeth (4 patients) were allocated in Class 1, 18 teeth (16 patients) in Class 2, 63 teeth (61 patients) in Class 3, and 16 teeth (13 patients) were allocated in Class 4. The success was defined as an "absence of resorption or signs of periapical or periodontal pathosis". Based on the results, Class 1 and Class 2 cases showed 100.0% success, Class 3 showed 77.8%, while Class 4 showed 12.5% success. The author emphasized the importance of an early diagnosis (159).

Other than the aforementioned cohort study (159), there is a plethora of case and case series reports (38,47,49,73,83,88–91,124–126,128–135,150–153,156,160–171) that provide information collected during the follow-up appointments, albeit this information is of low level of evidence; due to the type of these studies (172).

Chapter 2: Hypotheses and objectives

Three study hypotheses were postulated.

Study hypothesis 1: systemic determinants such as medication, systemic diseases, and allergies are associated with an increased prevalence of external cervical root resorptions (ECRR).

Study hypothesis 2: treatment-related determinants such as presence and type of external repair of ECRR, root canal treatment, resorption repair combined with root canal treatment are associated with lower rates of failure of teeth with ECRR.

Study hypothesis 3: local determinants such as location of tooth, location of portal of entry, Heithersay classification, trauma, orthodontic treatment, extraction of adjacent tooth, irregular forces and presence of restoration are associated with higher failure rates of teeth with ECRR.

In order to test the aforementioned hypotheses, the study objectives were as follows:

Objective 1: to identify all patients who were diagnosed with ECRR at the Graduate

Endodontics clinic of UBC, during a 10-year period of time (2008-2018).

Objective 2: to retrospectively review the medical charts of the ECRR patients, all information provided regarding their medical and dental history. Then, to analyze this information at a tooth and a patient level, and assess if it is associated with the presence of ECRR.

Objective 3: to follow-up all patients in the ECRR group and assess the long term treatment outcome and its potential determinants.

Chapter 3: Materials and methods

The current study included all patients who were referred to the UBC Graduate Endodontics (GE) clinic and who were diagnosed with ECRR, during the time period of 2008-2018. The inclusion criteria were: having a minimum of two periapical radiographs showing the defect, even taken at different times, or one periapical radiograph and one CBCT scan from the area of interest. The data were collected in two separate stages of the study. The first stage was a retrospective approach (retrospective chart review), where information was collected from the charts of patients with and without resorption. The group of patients without resorptions was the control group. The second stage was a prospective approach and included a follow-up of patients with ECRR. Inclusion criteria for the second stage was a follow-up interval not less than 1 year. The project was approved by the UBC Behavior Research Ethics Board (H17-00885).

3.1 Sample selection

Romexis software (Planmeca, Helsinki, Finland) was used to search for and identify patients with ECRR from the general pool of patients of the GE clinic. Romexis is the software suite used in the UBC faculty of dentistry, where all radiographs and other patient information, such as medical and dental history, are saved by chronological order in a patient's digital chart.

The first step in sample selection, was to use nine terms as key search terms that potentially can have been used in patients' digital charts. The following key terms were used: 1) resorp, 2) resorb, 3) resorption, 4) cervical resorption, 5) root resorption, 6) internal resorption, 7) internal root resorption, 8) external resorption, and 9) external root resorption. The objective of this step was to collect all patient charts that were including those key terms. Internal and cervical resorption can in some cases be difficult to differentiate from each other. Therefore, by including both internal and external resorption key terms, the possibility of not identifying an ECRR case because of previous misinterpretation of the resorption type was minimized After searching these nine terms in all GE patient charts (Fig. 6), two groups of information were collected. The first group of data showed the chart numbers of the cases where any of these key terms was used in the "Condition notes" section. That means that Graduate Endodontics residents used one or more of the terms while documenting their cases in the prelisted options in Romexis. The second group of data was showing if any of these key terms was used in "Free text notes", which means that residents used the key terms when adding comments in the free text notes area. Overall, 398 chart numbers were collected in the "Condition notes" group and a total of 308 in the "Free text notes" group. These chart numbers were combined and the duplicates excluded. This resulted in a total of 398 charts being collected from the first step of the sample selection. For the majority of these charts, the context in which that the key word was used was included and shown in the results, presented in an excel sheet (Fig. 7).



Figure 6. A print screen of a patient's chart in Romexis. The black arrow shows where one should "click" for the "search case history notes" option, which is a "search machine" for the key terms.

During the second step, 131 charts were excluded as irrelevant to the current research.

For instance, charts in which the term 'resorption' was used in a different context such as: bone resorption, sutures resorption or resorbable membrane were excluded (Fig.7). At the end of the second step of the sample selection, 267 charts with cases of RR or cases with unknown context of the key word were collected.

Но	me	Insert	Page	Layout Forr	nulas	Data	Review	View										
Past	, ∦ © ° ∢	Cut Copy + Format	Calib B	ri • 11 I <u>U</u> • 🔛	• A	A .	= =		°, • ⊒ •⊒	Wrap Text	General \$ * %	▼ 0.0,0.0.0 0.0 00.0 €	Conditional Formatting	Format as Table	Cell Styles	Insert	Delete	Form
A654		\times \checkmark	fx	98901														
	A	В		С							D							
OTO		20	107	100740000	CALCING	ai CCI VIC	aireaurp		ass 7									-
611		26-	+07	160746062														-
612		25-	+07	160/59906	reat re	tio	_											
614		25	+07	170259057	root re	root resultion												
615		20	+07	02626671	externa	ai root r	esorpion											#
012		20	-07	920200/1	interna	l root re	esorbtion											
616		25-	+07	100726890	incerna		coor belor											#
617		26	+07	180803754	tooth h	nas a res	orbed er	ndodont	tically t	x root								
618		26-	+07	104756288	apical r	esorpti	on.											#
619		2E-	+07	161747700	inflam	natory	root reso	rption										1
620		2 2E-	+07	84834681	resorpt	tion of a	pical bor	ne .										
621		7 2E-	+07	111451306	internal and external resorption #													
622		2 2E-	+07	130834290	resorpt	tion												#
					either r	root res	orption o	r iatrog	enic da	mage during surgica	extraction of	of 17 (more c	onsistent v	vith trac	uma/iatr	ogenic tr	auma	
623		7 2E-	+07	121152540	during	extracti	on of mo	lar)										#
624		2E-	+07	121152650														

Figure 7. Excel sheet illustrates the organization of the information given about the key terms found in Romexis "Free text notes". Note that in some cases the context and/or the tooth number is provided.

During the third step of the sample selection, the 267 charts were thoroughly examined. Each patient was given a code number in order to be identifiable only by the research team. At this stage, 98 charts were excluded as irrelevant to the current project. Then the RR group was divided into two subcategories, 1) resorption at the cervical and/or middle root third (Fig. 8a) and 2) resorption at the apical third (Fig. 8b). Seventy-eight cases with an apical resorption were excluded as irrelevant to ECRR. At the end of the third step of selection, the remaining sample was 91 patients with radiolucency at the cervical or middle third of a root of at least one tooth.



Figure 8. Examples of different locations of root resorption (pointed by a yellow arrow): a) cervical root third b) apical root third.

During the fourth step, radiographs of 91 patients were exported from the Romexis digital patients' charts, and organized per patient and per date taken. Multiple periapical radiographs of each patient, and panoramic or CBCT scans, if available, were imported into a Power Point presentation (Microsoft® PowerPoint). A black background was used for the slides. Every time a new case was to be presented, the first slide was a total black slide with only a patient's code number, making the transition clear. In the second slide of each patient, periapical radiographs were added with information of the year these radiographs were taken. If panoramic radiographs or CBCT scans were available, they were presented on the following slides. This PowerPointbased information was assessed twice, six months apart. The first time it was presented on a 13-inch MacBook Air with a screen resolution of 2560 by 1600 pixels. The second time, six months later, the radiographs in the PowerPoint presentation were viewed on a 40-inch Samsung monitor. Two examiners (one supervisor and the MSc candidate) assessed all cases. In the assessment a decision was made on the following parameters: location of the RR (internal/external) and the Heithersay classification. In cases where there were doubts or

disagreement, the whole CBCT scan, and not only a print screen, was examined. If a CBCT was not available, the clinical description provided in a patient's chart was used. Due to the limitation of Heithersay classification, which is based on a two dimensional scale, the examiners came to an agreement on how to classify a specific type of lesion. More specifically, the cases that were *"well-defined"* (part of the definition of Class 2) and at the same time *"extending at least at the coronal third of the root"* (part of the definition of Class 3). The examiners concluded that all cases that had well-defined borders, without the typical channeling of Classes 3 and 4 around the root canal, should be defined as Class 2 (Fig. 9).



Figure 9. Example of Class 2 cases with well-defined borders and without a channeling spread: a) periapical radiograph and b) CBCT axial, coronal, and sagittal views of the same Class 2 case.

At this stage, 16 cases were excluded as internal inflammatory RR cases. Another four cases with radiolucent defects on the cervical or middle third of the root were also excluded. One

case was excluded because there was only one periapical radiograph in a patient's chart, and the second one because the resorptive defect was present only in one of the periapical radiographs (Fig. 10a), leading to a possible false positive diagnosis of ECRR (Fig. 10b). The third and the fourth cases were excluded because they were obvious incorrect diagnoses, not ECRR, in contrast to the clinical notes. Finally, by the end of the summer of 2017 when all data were collected, the group of patients diagnosed with ECRR at the UBC GE clinic consisted of 71 patients and 92 teeth.



Figure 10. An excluded case, showing an example of false positive diagnosis of external cervical root resorption (yellow arrows point the tooth of interest): a and b periapical radiographs were taken the same day.

The fifth step was to keep the sample updated while this project was still ongoing. For that reason, five recently referred patients, from September 2017 till May 2018, to the UBC GE clinic and newly diagnosed with ECRR, were also included in the sample.

After these five steps were followed (Fig. 11), a total of 76 patients and 98 teeth diagnosed with ECRR were included in the sample.

	Sample selection
1st step Key terms applied in all Romexis GE patient charts	398 patients having a key term in their chart
2nd step Chart exclusion based on the context that key words were used	398 patients - <u>131 irrelevant charts</u> 267 patients
3rd step 267 charts thoroughly examined. Differentiation between apical and cervical/middle location of radiolucency	267 patients - 98 irrelevant charts 78 patients with apical RR 91 patients with cervical/middle RR
4 th step Differentiation between internal and external radiolucency by 2 examiners	Radiographic assessment by 2 examiners 91 patients with cervical/middle RR - 16 patients with internal RR <u>4 patients did not meet inclusion crite</u> ria 71 patients with ECRR identified in Romexis
5 th step Keep sample updated	71 patients with ECRR identified in Romexis + 5 new patients with ECRR 76 patients with ECRR

Figure 11. Chart illustrates the 5 steps followed for the sample selection of the external cervical root resorption group (ECRR).

Finally, after the ECRR group was identified, another group of patients (control group), with an equal number of participants, was created. The inclusion criteria for this group were exactly the same as for the study group: first a participant had to be chosen from the pool of patients of GE clinic and secondly the patient had to have two radiographs from the area of concern. However, none of the patients included in the control group had ECRR. During this selection process, a total of 76 patients without ECRR, each with two periapical radiographs were included in the control group.

3.2 Data collection

Data were collected at two different stages of the study for two different purposes. In the first stage, the MSc candidate retrospectively collected information from the patients' charts when they were first diagnosed with ECRR. The second stage of data collection (prospective review) included information of the most recent follow-up appointment, which had to be at least one year after the initial diagnosis.

After the group of patients with ECRR was identified, data from their charts were collected and recorded on an Excel sheet (Microsoft® Excel). The following retrospective data was collected:

- 1. year of birth
- 2. gender
- 3. detailed information regarding their medical history (list of 21 medical conditions)
- 4. medication use
- 5. allergies
- 6. number of a tooth diagnosed with ECRR
- 7. year of diagnosis of ECRR
- 8. potential predisposing factors: trauma, orthodontics, extraction of adjacent tooth restoration (direct, indirect, post) and irregular forces (bruxism, eruption disorders)
- 9. patient's chief complaint
- 10. if two periapical radiographs were available
- 11. if CBCT scan of the area of interest was available
- 12. radiographic findings indicating where the resorption was located (mesial, distal, palatal/lingual, buccal)
- 13. class of ECRR based on the Heithersay classification

14. pulpal diagnosis

- 15. periapical diagnosis
- 16. initial ECRR repair/treatment: no treatment, internal approach, external approach, extraction, root amputation
- 17. the material(s) used for the resorption repair

The same information, except for the items 6, 7, 12, 13, 16 and 17, were retrospectively collected for the control group.

After all information was collected, patients identified with ECRR were contacted, in order to arrange a follow-up appointment. The MSc candidate tried to contact all patients who were diagnosed with ECRR at least one year earlier. Patients who were recently diagnosed with ECRR and did not meet the criterion of a minimum one-year follow-up, were not contacted. However, these patients remained in the patients' list for future follow-up. In addition, if a recent follow-up had taken place during the last year (173, 174), these patients were not contacted, but the ECRR related information was retrieved from their Romexis digital charts. The MSc candidate tried to contact all eligible participants via phone, initial contact letter and an initial contact email in order to arrange a follow-up appointment. For the patients that were willing to participate but were living in remote areas, thus not able to visit the UBC clinic, a different solution was found. Initially, a verbal permission was acquired from these patients, followed by their dentist's contact information. The next step was to contact their dentists and ask them to provide chart information and copies of radiographs of the area of interest. In cases where dentists did not have radiographs of a tooth with ECRR or if patients did not have a dentist or they were unwilling to provide their contact information, a quick interview over the phone took place. During this

interview, the patients or their dentists (without periapical radiographs but with clinical charts), were asked to answer the following questions: 1. Is a tooth (with ECRR) still in place? 2. If extracted, when and why did that happen? 3. If a tooth is still present, does a patient have any symptoms or clinical findings? 4. Could you describe the chief complain?

Compensation was offered to all patients for the lost working hours and travel expenses. To partially compensate participants, each of them received a 10 CAD Save on Foods groceries gift card. The follow-up appointments were free of charge, based on the protocol of the UBC GE clinic.

Follow-up appointment and collection of data was not conducted for the control group, because the assessment of endodontic treatment outcome was out of the scope of the present study. During the follow-up appointments, the protocol of the UBC GE clinic was followed. First, all participants signed an informed consent before the examination took place. During the clinical examination of the tooth of interest and its adjacent teeth; chief complaint, probing, palpation test, percussion test, bite test, vitality test were conducted and intraoral photos were taken by using a digital camera and the dental operating microscope. During the radiographic examination, periapical radiographs were taken, and if needed for treatment purposes, a CBCT scan was taken as well.

After several efforts were made to contact all patients for a recent follow-up, the following data was collected:

- 1. year of the last follow-up
- 2. if there was no follow-up, why
- 3. clinical findings /chief complaint
- 4. radiolucency in bone, related to ECRR
- 5. presence of RCT at the final follow-up

- 6. year of extraction
- 7. reason for the extraction
- 8. change of size of the ECRR
- 9. failed or functional tooth.

For optimal evaluation of the change in size of the ECRR, different CBCT scans should ideally be used. The digital subtraction radiography technique has been used, in a case report, in order to compare the size of the ECRR radiolucency in the initial and the final CBCT slices (131). As mentioned earlier, CBCT is better suited for the evaluation of the size of ECRR lesions than conventional 2D radiographs. However, not all of the patients in our sample had a CBCT scan. Due to this limitation, the initial and the latest periapical radiographs of each case were used. The radiographs were presented on a Power Point presentation after all the follow-up appointments were completed. For each case, a single slide with black background was used to present the initial and the latest periapical radiographs. If the size of ECRR defect changed during the follow-up period in untreated cases, or a new radiolucency was detected around the ECRR restoration in treated cases, this information was recorded. No other comment on the extension of the ECRR was made, because it was not possible to make quantitative measurements of the changes (Fig. 12).





Case 1: a) initial periapical radiograph taken in 2013 and b) latest periapical radiograph taken in 2018. Note the increase in the ECRR size.

Case 2: c) initial periapical radiograph taken in 2016, and d) latest periapical radiograph taken in 2018. The ECRR lesion seems unchanged in the 2D radiographs.

For the outcome assessment, it was critical to define the difference between a functional and a failed tooth. For that reason, Friedman's classification for treatment outcome was used, and a functional tooth was defined when " *the clinical presentation was normal, while radiolucency might be absent or present – newly emerged or persisting*" (175–177). Accordingly, as failed was defined a tooth when it was extracted or it was a non-functional (tooth without normal clinical presentation).

3.3 Data analyses

The data was analyzed with the SPSS version 25.0 software (IBM, New York, United States) and the level for statistical significance was set at p<0.05. Data were analyzed at two levels: patient as a unit of analysis and tooth as a unit of analysis. The analysis of patient level data used comparisons between the patients with external cervical root resorption (study group) and the patients without external cervical root resorption (control group). For both type of statistical comparisons (patient- or tooth-based data) Chi Square test or Fisher's Exact test were used to compare proportions. To get insight into time-related failure rates, survival analyses employing Kaplan Meier curves method was used, where different local treatment-related and predisposing risk factors (determinants) were considered.

Chapter 4: Results

4.1 Patient level of analysis

In this section, the collected data and the analysis regarding all the patients diagnosed with ECRR, irrespective of the existence of follow-up information, is reported. The results on a patient level will be presented in three different approaches:

1) presentation of retrospectively collected information from patients' charts, related to their medical and dental history. The results are visually illustrated in Figure 13.

2) comparison between the retrospectively collected findings of ECRR and the control groups. The results are visually illustrated in Figures 14 and 15, and Table 1.

3) presentation of the time intervals and the number of patients who were followed-up. The results are visually illustrated in Figures 16 and 17.

Univariate descriptive analyses: Patient level of analysis, Retrospective approach

This first approach analyzed the data at a patient level; based on the list of retrospectively collected information presented previously. Regarding the prevalence of ECRR, 2.3% of all patients at GE clinic were diagnosed with at least one tooth with ECRR. The sample age-related distribution (age of patient when initially diagnosed with ECRR) was as follows: 12-39 years (n=12, 15.3%), 40-59 years (n=28, 36.7%) and 60-89 years (n=27, 35.7%) (Fig. 13). Of the 76 patients, 31 were women and 45 were men. Fifteen patients were diagnosed with more than one tooth with ECRR (ten patients with two teeth, four patients with three teeth and one patient with five teeth with ECRR). Twenty-four patients had a non-contributory medical history (no systemic condition), while 52 patients had at least one systemic disease, which will be discussed in further detail. Nineteen patients had a complex medication regimen, while 57 took either no

medications or light medications such as vitamins or painkillers only when needed. Fifty-one patients did not report any allergies, while 16 patients had allergies. Regarding the initial patient's chief complaint, 48 patients had none, 24 expressed some pain or sensitivity, while four patients were concerned about the aesthetics of a tooth due to a pink discoloration. All patients finally included in the study had at least two periapical radiographs of the tooth of interest, and 46 patients also had a CBCT scan from the area of concern.



Distribution of the age of patients

Figure 13. Distribution of the age of patients at the time of external cervical root resorption (ECRR) diagnosis.

Bivariate analyses: Patient level of analysis, Case-control study approach

The second approach provides information regarding the medical and dental history of the ECRR group and of the control group. Chi square test was conducted to compare proportions. The results of the systemic determinants are presented in Table 1.

Overall, the patients in the control group had more systemic conditions than the patients in the ECRR group (Table 1). More specifically, a higher number of patients in the control group, compared to the patients of the ECRR group, presented with a medical history related to musculoskeletal disorder, high blood pressure, orofacial pain, gastrointestinal disorder, heart vascular disease, thyroid disorder, CNS disorder, hemorrhagic disease, dermatological problems, chemotherapy, eating disorders, presence of allergies and need for medication. However, the differences regarding the presence of the aforementioned conditions among the two groups were not statistically significant. In addition, for some medical conditions, the number of patients presenting them were identical between the two groups. More specifically, the disorders with equal proportions between the groups were: kidney disease, liver disease, mental health problem, sleep disorder, rheumatoid disorder, organ transplantation and dry mouth. There were four systemic conditions which presented the opposite trend and were observed more frequently in the ECRR group than in the control group. Three of them, respiratory disease, head/neck radiotherapy and implants (not dental) did not present significant differences between the control and ECRR group. The presence of diabetes was the only statistically significant difference between the two groups.

In summary, there was a tendency for more systemic health conditions in the control group, but no significant differences between the two groups were detected except for diabetes. The number of diabetics in the ECRR group was three times higher when compared to the control group.
Table 1. Potential determinants: presence of systemic conditions

Systemic condition	Control group	ECRR group	Significance #
	n (%)	n (%)	(p-value)
Medical condition (any)	58 (76.3)	52 (68.4)	0.276
Musculoskeletal	26 (34)	16 (21.1)	0.070
High blood pressure	23 (30.3)	21 (27.6)	0.721
Respiratory disease	6 (7.9)	9 (11.8)	0.415
Kidney disease	4 (5.3)	4 (5.3)	1.000
Liver disease	3 (3.9)	3 (3.9)	1.000
Mental health	12 (15.8)	12 (15.8)	1.000
Orofacial pain	4 (5.3)	3 (3.9)	0.699
Gastrointestinal	16 (21.1)	8 (10.5)	0.075
Sleep disorder	12 (15.8)	12 (15.8)	1.000
Heart vascular disease	16 (21.1)	15 (19.7)	0.840
Thyroid	12 (15.8)	10 (13.2)	0.645
Central nervous system	7 (9.2)	5 (6.6)	0.547
Hemorrhagic disease	4 (5.3)	3 (3.9)	0.699
Diabetes	4 (5.3)	12 (15.8)	0.034
Dermatology	13 (17.1)	11 (14.5)	0.656
Rheumatoid disorder	4 (5.3)	4 (5.3)	1.000

Chemotherapy	4 (5.3)	1 (1.3)	0.172
Head/neck radiotherapy	0 (0.0)	3 (3.9)	0.080
Organ transplantation	1 (1.3)	1 (1.3)	1.000
Dry mouth	10 (13.2)	10 (13.2)	1.000
Implants (non-dental)	4 (5.3)	5 (6.6)	0.731
Eating disorders	4 (5.3)	3 (3.9)	0.699
Allergies	28 (36.8)	25 (32.9)	0.610
Medication	26 (34.2)	19 (25.0)	0.214

#Chi-Square Test

Comparison of local risk factors revealed that the ECRR group presented more patients with a history of trauma and less patients with any kind of restoration, when compared to the control group. Other local determinants presented similar numbers for both groups (Fig. 14).

Number of patients with potential local predisposing factors



Figure 14. Bar chart illustrates the number of patients presenting different potential local predisposing factors in the control and the external cervical root resorption (ECRR) groups.

A single local factor rather than several factors was more frequently present, 48.0% in the ECRR and 67.0% in the control group (Fig. 15). Also, both groups had comparable proportions for the presence of two local factors, with 29.0% for the ECRR group and 24.0% for the control group. The ECRR group had a few cases with 3 or 4 local determinants (1.0%) (Fig. 15).



Simultaneously present potential local predisposing factors

Figure 15. Pie charts illustrate the frequency of simultaneous presence of different potential local predisposing factors in the control and external cervical root resorption (ECRR) groups.

Univariate descriptive analyses: Patient level of analysis, Prospective approach – Follow-up

The third approach to analysis provides information regarding the number of patients who were examined at one or more follow-up appointments. The mean follow-up time was 3.9 years (SD=3.1 years, minimum 6 months and maximum 10 years). Out of the 76 patients with ECRR, 20 patients were not recently followed-up, because five were newly diagnosed, 14 had relocated and could not be contacted or did not respond, and one was deceased. Overall, 73.7% of the patients with ECRR were recently followed-up. Out of the 56 recent follow-up appointments, 23 took place at the UBC faculty of dentistry, 8 were conducted by other dentists who provided radiographic and clinical information, and 8 were interviewed by phone. From the

overall group, 17 patients had their teeth extracted. The extraction was confirmed either by information provided in the patients' chart (n=15 patients) or during the phone interview (n=2 patients). Two patients with more than one teeth with ECRR had one of them extracted (Fig. 16).



5 newly diagnosed with ECRR patients

Figure 16. Chart illustrating the number of patients who were or were not recently examined at a follow-up appointment.

Eleven out of the 20 patients that we were unable to follow-up within the last year, had a follow-up information provided in their charts. This information was used as their last follow-up. This way, the overall follow-up information was available for 67 patients (88.2% follow-up rate). From the nine patients that were never re-examined in a follow-up appointment, five were newly diagnosed with ECRR, three had relocated and therefore could not be contacted and one was terminally ill at the time of the original ECRR diagnosis (Fig. 17).



Figure 17. Chart illustrating the number of patients who were or were not examined at a follow-up appointment.

4.2 Tooth level of analysis

In this section, the unit of analysis was the tooth. In these analyses, control group patients were not included. The results on the tooth level will be presented in four different approaches: 1) presentation of the retrospectively collected data of all teeth diagnosed with ECRR. The results are visually illustrated in Figures 18 - 21.

2) presentation of findings collected at the follow up appointments (prospective approach).

3) presentation of the survival rates of teeth having at least a one-year follow-up, based on different local determinants. The results are interpreted in Tables 2 and 3. Bivariate analysis was used to present this information.

4) presentation of the survival rates of teeth having at least a one-year follow-up, based on different local determinants. The visual analysis of failure rates is illustrated in Figures 23 - 30. Hazard ratio curves were used to present this information.

For the last two approaches, the survival rates of teeth from different ECRR subgroups (regarding the presence of local determinants) were compared to each other.

Univariate descriptive analyses: Tooth level of analysis, Retrospective approach

Fifty-six of the 98 teeth with ECRR were anterior teeth (31 maxillary and 25 mandibular), and 42 were posterior teeth (20 maxillary and 22 mandibular). The distribution of the teeth is presented in Figure 18.



Figure 18. Distribution of teeth with external cervical root resorption.

There was an almost equal distribution among teeth with "normal pulp" (n=29 teeth, 29.5%), "pulp necrosis" (n=33 teeth, 33.7%) and "root filled teeth" (n=28 teeth, 28.6%). The diagnosis of "irreversible pulpitis" (symptomatic or asymptomatic) was less frequent (n=7 teeth, 7.1%), and the least frequent diagnosis was the "previously initiated RCT" (n=1 tooth, 1.1%).

The majority of the teeth, 68 out of 98 teeth (69.4%), were symptom free at the time of the diagnosis. The "normal apical tissues" was the most predominant finding (n=48 teeth, 49.0%); followed by an "asymptomatic apical periodontitis" (n=20 teeth, 20.4%). Slightly less frequent than the asymptomatic assigned finding, was the "symptomatic apical periodontitis" (n=16 teeth, 16.3%), and even less common was the "chronic apical abscess" (n=11 teeth, 11.2%) finding. The presence of "acute apical abscess", presented an opposite trend from the predominant asymptomatic conditions (n=3 teeth, 3.1%).

The POE was located on the buccal surface of the tooth in 25 cases (25.5%), on the palatal or lingual surface in 23 cases (23.5%), on the mesial side in 26 cases (26.5%) and on the distal side on 21 cases (21.4%). In 3 cases (3.1%) the location of the POE was not identifiable because the extend of the resorptive channels was 360°, circumferentially around the root canal space. In 5 cases (5.1%) the resorptive defect covered 3 surfaces of the root and in 16 cases (16.3%) covered 2 surfaces. Seventy-four cases (75.5%), the majority of the resorptive defects, had a single surface presentation. When the location of the POE was not identifiable, the center of the resorptive defect was noted instead (Fig. 19).



Figure 19. Distribution of teeth based on the location of the portal of entry (POE) or, when it is not visible, the center of the resorptive defect.

The frequencies of the different Heithersay ECRR classes 1, 2, 3 and 4 were 12.2% (n=12 teeth), 38.8% (n=38 teeth), 23.5% (n=23 teeth) and 25.5% (n=25 teeth) respectively (Fig. 20).



Figure 20. Distribution of teeth based on the Heithersay classification.

The sample was divided almost equally in numbers of teeth that received no resorption repair (49.0%) and teeth that received some kind of resorption treatment (51.0%). More specifically, 24 teeth (24.5%) received internal repair and in 17 teeth (17.3%) the resorptive process was accessed from outside (external approach, directly from external root surface). Six teeth with ECRR (6.1%) were extracted, and in three multi-rooted teeth (3.1%) amputation of one root was performed. For the internally repaired teeth, the following materials were used: MTA in 17 teeth (70.1%), resin composite in three teeth (13.4%), Biodentine in two teeth (8.3%), GI in one tooth (4.1%), and unknown material in one tooth (4.1%). For the externally repaired ECRR defects, resin composite was used in ten teeth (59.0%), Geristore in four teeth (23.6%), and MTA (5.8%), GI (5.8%) and amalgam (5.8%) on one tooth each. Overall, MTA was the most frequently used material for the internal repair and resin composite for the external repair (Fig. 21).

Initial ECCR treatment management

(tooth level of analysis)



Figure 21. Chart illustrating the number of teeth which did or not receive initial treatment of the external cervical root resorption (ECRR). The materials used for the internal and the external repair and their frequencies are also pictured.

Univariate analyses: Tooth level analysis, Prospective approach

Survival rates were calculated for all teeth with at least one-year follow-up (n=89 teeth). Thirteen teeth developed new or increased periodontal radiolucency, five of them were related to the ECRR area and eight to periapical area. Five teeth presented with increased ECRR size, based on the radiographic findings, all of them without ECRR repair or RCT. Overall, 24 teeth were categorized as failed, and 19 of them were already extracted at the time of the follow-up procedure. The remaining five teeth were categorized as not functional based on the follow-up findings, even if they were still present in the oral cavity. Out of the 19 extracted already teeth, four were extracted due to periodontal reasons, eight because they were not restorable, and seven due to a poor prognosis. Out of the five still present but failed teeth, two failed due to periodontal

problems (Fig. 22), two because they cracked, and one due to vertical root fracture which was not related to the resorption side. All above categorizations, for instance the three etiological factors for extraction (i.e. periodontal problems, restorability and poor prognosis), are organized based on the clinical notes or phone interview findings.



Figure 22. Clinical pictures of a tooth (pointed by a black arrow) failed due to a periodontal problem. a) intraoral picture of tooth #43 taken during the follow-up. b, c and d) intraoral pictures taken three months later than picture a. Pictures b and c show the periodontal inflammation and d) shows the tooth before the completion of the extraction. Note the extend of external cervical root resorption, and the root canal space in the center of the resorptive defect (yellow arrow). For the radiographic interpretation of the case, see Figure 12 a and b. Bivariate analyses: Tooth level analysis, Failure rates regarding different local determinants

Two treatment related local factors showed statistically significant differences regarding the ECRR failure rates. The first factor was the RCT, and it showed that root canal treated teeth had a lower risk of failure than the teeth without such treatment. The second factor was the ECRR repair, whether combined or not with RCT. The comparison showed that teeth with ECRR repair and RCT had lower failure rates than teeth with ECRR repair but no RCT. Overall, the presence of ECRR repair and whether internal or external approach was used, showed no statistically significant difference in the ECRR failure rates (Table 2).

Treatment related determinants	Failure rates n (%)	Significance # (p-value)
Repair of External Root resorption		
Internal	3 (13.6)	0.372
External	4 (25.0)	
Root canal treatment		
Yes	9 (17.3)	0.007
No	15 (44.1)	
Resorption repair & Root canal treatment		
Yes	6 (15.0)	0.013
No	18 (39.1)	
Initial root resorption repair		
Yes	11 (23.4)	0.307
No	13 (33.3)	

Table 2. Treatment related determinants in ass	sociation with failure rates (tooth level analysis)
--	---

Chi Square test

The "location of the portal of entry", "history of trauma", "orthodontic treatment", "extraction of the adjacent tooth" and the "presence of restoration" were local determinants without statistically significant difference in the ECRR failure rates. The factor "irregular forces" was statistically significantly related to the failure rates (p=0.032). However, the 75% of teeth with irregular forces that failed, represents only three teeth. Because of the small number of teeth, this finding cannot be considered as truly significant.

Two local determinants, tooth location and Heithersay class, showed statistically significant differences regarding the ECRR failure rates: posterior teeth with ECRR had higher failure rates compared to anterior teeth, and the higher the Heithersay class, the higher the risk of failure (Table 3).

Local determinants	Failure rates n (%)	Significance # (p-value)
Location (tooth)		
Anterior	8 (16.7)	0.009
Posterior	16 (42.1)	
Location (portal of entry)		
Buccal	7 (31.8)	
Distal	4 (21.1)	
Palatal/lingual	4 (21.1)	0.802
Mesial	8 (34.8)	
360	1 (33.3)	

Table 3. Local determinants in association with failure rates (tooth level analysis)

Classification		
Class 1	0 (0.0)	0.020
Class 2	6 (18.2)	
Class 3	7 (33.3)	
Class 4	11 (47.8)	
Trauma		
Yes	4 (16.7)	0.148
No	20 (32.3)	
Orthodontic treatment		
Yes	1 (20.0)	0.685
No	23 (28.4)	
Extraction of adjacent tooth		
Yes	5 (33.3)	0.606
No	19 (26.8)	
Restoration		
Yes	12 (26.1)	0.526
No	12 (32.4)	
Irregular forces		
Yes	3 (75.0)	0.032
No	21 (25.6)	

Chi Square test

Visual analyses of failure rates: Tooth level analysis, hazard ratio curves

The cumulative hazard, meaning the cumulative probability of failure in different followup years, is demonstrated in a subsequent series of graphs. Within the first year there was a 12.0% probability of failure (Fig. 23). Corresponding failure probabilities were for the 2nd, 3rd ,4th and 5th years were 14.0%, 16.0%, 23.0% and 26.0% accordingly. At the 6th year a substantial increase in failure rates was observed, when half of the cases showed an increased probability of failure at 8 years, which could otherwise be interpreted that half of the cases survived 8 years. The highest 80.0% cumulative failure rate was at the 10th follow-up year (Fig. 23).



Figure 23. Treatment outcome and probability during a failure of a 10-year follow-up period.

Risk probability (cumulative hazard) illustrated in Figure 24 compares cases with no ECRR repair to the cases with some ECRR treatment (internal or external approach, root amputation or immediate recommended extraction). The probability of risk was relatively similar for both groups throughout the first 7 years. For the last 2 years, the probability of failure was high in both treatment groups (Fig. 24).



Figure 24. Treatment outcome. Determinant: external cervical root resorption (ECRR) repair.

The failure rates were compared between the teeth which were restored either via internal or external access (Fig. 25). For the first four years there were no obvious differences between the two groups of repair. However, after the 7th year the external repair group showed an increased risk of failure. The internal repair did not show any increased risk after the first four years.



Figure 25. Treatment outcome. Determinant: Type of repair of external cervical root resorption repair.

When risk rates were compared between different locations (Fig. 26), the anterior teeth showed lower probability of failure than the posterior teeth. More specifically, the probability of failure was below 30.0% for the anterior teeth up to the 8th to 10th years of follow-up, which was similar to failure rates of posterior teeth at the 4th and 5th year of follow-ups. For the posterior teeth, the probability of failure was around 90.0% at the 7th to 10th year of follow-ups.



Figure 26. Treatment outcome. Determinant: Location of the tooth, anterior or posterior.

The treatment outcomes were also evaluated based on tooth vitality (Fig. 27) by comparing risk rates between teeth that received and those that did not receive root canal treatment (RCT). Overall, the probability of failure was higher for the teeth without RCT, e.g. at the 2nd year of follow-up the probability of failure was 35.0% for the teeth without RCT, while it was 5.0% for the teeth with RCT. The risk of failure was 70.0% for the teeth without RCT and 40.0% with RCT at the 8th year of follow-up.



Figure 27. Treatment outcome. Determinant: Root canal treatment (RCT).

The probability of failure was also examined based on resorption repair with or without RCT (Fig. 28). Overall, the teeth without RCT but with repaired ECRR showed a substantially higher probability of failure, i.e. the risk of failure at the 5th year was 35.0% vs. 15.0%, while for the 8th year it is 65.0% and 35.0% respectively, for the teeth without RCT teeth and for the teeth with RCT (Fig. 28).



Figure 28. Treatment outcome. Determinant: External cervical root resorption repair combined with or without root canal treatment (RCT).

The failure rates were also assessed based on the location of the portal of entry (POE) or, if it was not visible, the center of the resorptive lesion as well as the number of tooth surface affected (Fig. 29). When the resorption spread was circumferential (360° around the root canal system), the risk of failure was substantial and reached 70.0% at the 1st year. At the end of the follow-up period, the risk of failure was higher when the POE was located on an interproximal surface compared to the location at the buccal and palatal/lingual surface.



Figure 29. Treatment outcome. Determinant: Number of surface affected, location of portal of entry (POE), or if not visible, center of resorptive defect.

The failure rates in teeth with a different Heithersay ECRR class were compared (Fig. 30). Class 4 teeth showed the highest risk of failure, starting with 25.0% the 1st year, and reaching the 100.0% the 5th year. Class 1 showed 0.0% risk of failure throughout the whole 10year follow-up period. At the 5th year, class 3 teeth showed a 5.0% risk of failure in comparison to 20.0% for class 2. After the 7th year the risk pattern changed. At the 10th year, class 3 teeth had 70.0% and class 2 had 40.0% failure rates (Fig. 30).



Figure 30. Treatment outcome. Determinant: Heithersay classification.

Chapter 5: Discussion

5.1 Discussion

First study hypothesis

The first study hypothesis was that systemic determinants are associated with higher rates of presence of external cervical root resorptions. To the best of our knowledge, this hypothesis has not been examined in previous studies. Systemic factors included presence of allergies, use of medication and 21 different medical conditions were assessed with a case-control study approach. Although the control group patients were unhealthier overall, there were no statistically significant differences between the study group (external cervical root resorption) and the control group (no external root resorption), except for one systemic condition. Diabetes was statistically significantly more frequent in the ECRR patients than in the control group. This finding indicates that diabetes may be a risk factor for ECRR, which partially supports the first study hypothesis. A potential link between ECRR and diabetes, could be hypoxia as oxidative stress and inflammation seem to be related to hypoxic cellular microenvironment in diabetic patients (178). In other studies, hypoxia has been proposed to predominate during the first steps of ECRR (53,54).

There are no previous studies assessing the systemic determinants of ECRR, however, there are two cohort studies regarding the potential local predisposing factors related to ECRR (3,6). In these two studies and the current study, the gender distribution was similar with the percentage of women diagnosed with ECRR being 40.8% (present study), 48.0% (6) and 51.3% (3). However, there was a difference in the age distribution at the time of the diagnosis of ECRR. In our study, the majority of cases was almost equally distributed in two age groups: 40-59 years

78

(n=28, 36.7%) and 60-89 years (n=27, 35.7%), whereas in the study of Mavridou et al. (6) the majority of the patients were between 15 and 54 years, and in Hethersay's study between 20 and 50 years (3). The age of patients in our study sample was higher at the time of the diagnosis of the ECRR than in the other two cohort studies. This age difference could be due to the relatively small number of patients in our sample (76 patients), in comparison to the other two studies, which had 284 (6) and 222 patients (3). All three studies showed that the maxillary central incisors were the most commonly affected teeth. The prevalence of ECRR in our study was 2.3%, which was calculated as the percentage that was diagnosed with ECRR from all patients in the graduate endodontic program during the last 10 years. In Heithersay's study (3), the ECRR prevalence was 0.02%, and it was calculated as the percentage of people living in Adelaide (Australia) in 1999 (1.2 million people) that were diagnosed with ECRR. The difference in the prevalence could be explained by that fact that Heithersay (3) used a significantly larger population (Adelaide population) than in our study for the calculations.

Second study hypothesis

The second hypothesis of the present study was that treatment-related determinants are associated with lower rates of failure. Based on the results of the present study, no statistically significant difference between the internal and the external repair of the ECRR was found. Two statistically significant differences regarding the ECRR treatment outcome were identified: "ECRR repair combined with or without RCT", and "RCT present vs. absent". Time interval for the presence of the RCT is the latest appointment recorded in a patient's chart, not the initial diagnosis. The two statistically significant findings were the following: first, teeth with ECRR repair combined with RCT, and second, root canal treated teeth regardless of the presence of

ECRR repair, showed lower failure rates. These two factors had Kaplan Meier hazard curves with similar patterns. When the determinant was "ECRR repair or not", there was no statistically significant difference between the "ECRR repair" and the "no repair group". This is a finding of great clinical interest as the result indicates that the "no treatment" potentially could be a valid treatment option for some ECRR cases. However, it should be mentioned that in our sample, the five teeth presenting with increased size of ECRR, were all untreated cases. Overall, the second hypothesis was partially accepted, since some ECRR treatment-related factors, but not all, were associated with lower failure rates. To the best of our knowledge, there are no published studies which have examined the relationship of ECRR treatment-related factors and treatment outcome.

Third study hypothesis

The third hypothesis was that local determinants are associated with higher failure rates. No statistically significant differences in ECRR treatment outcome were found when the determinants were evaluated: "location of portal of entry or center of resorptive defect", "trauma", "orthodontic treatment", "extraction of adjacent tooth" and "restoration". When the "irregular forces" was examined as a determinant, it was statistically significantly associated with the treatment outcome (75.0% failure rate). However, one must pay attention on how this information is interpreted. This high percentage of failures (75.0%) represents three failed cases out of total four cases with "irregular forces". Due to the small sample size of the present study, potential misleading outcomes should be assessed in depth before confirming this as valid evidence. A statistically significant difference regarding the treatment outcome was found when the determinant "tooth location" was evaluated. More specifically, the failure rate was significantly higher for the posterior teeth than the anterior teeth. The difference in failures

80

increased substantially after the 6th year. This could potentially be related to the higher frequency of cracks on posterior teeth (179). To the best of our knowledge, there are no previously published studies which have examined the aforementioned local predisposing risk factors as determinants for the ECRR treatment outcome. However, there is one published study, which examined the treatment outcome based on the Heithersay classification as a local predisposing factor (159). Some similarities and differences between our study and the aforementioned one (159) will be discussed; prior to the results being presented. Both studies had similar size samples: 89 teeth in our study and 101 in the Heithersay study. The minimum follow-up period was one year in our study and three years in the Heithersay study (159). We defined the teeth as functional when there were no clinical findings, regardless of radiographic appearance, while in the other study, success was defined as the absence of resorption and other pathological findings in the radiographs. Our sample included both treated and nontreated teeth, where treatments were provided by different clinicians with different treatment approaches (e.g. multiple different materials were used). On the other hand, the Heithersay's study (159) included only treated cases, treated by one clinician (the author) who followed a specific protocol every time. In our study, a statistically significant difference regarding the treatment outcome was found when the determinant was the "Heithersay classification". More specifically, the failure rate was 0.0% for class 1 cases, and 18.2% for class 2 in our study, while Heithersay presented 100.0% success for both class 1 and class 2 cases. In our study, the failure rate for class 3 cases was 33.3% while in Heithersay's study the failure rate was 22.2%. For class 4 cases the failure rates were 47.8% and 87.5%, respectively. Overall, the third hypothesis is partially accepted, since some but not all potential local predisposing factors were associated with higher failure rates of teeth with ECRR.

An important difference was identified on how the studies by Heithersay (3) and Mavridou et al. (6) were conducted in comparison to the present study. In our study, all data were retrospectively collected from patient charts, whereas in the other two studies the authors collected information by directly interviewing patients. Because of that, the number of local factors we were able to assess was predetermined. For instance, playing of wind music instrument has been described as local predisposing factor in one of those cohort studies (6) and in a case report (95). The fact that in our patients' charts there was no information provided about use of wind music instruments, should not be conclusive evidence of the absence of this predisposing factor. As a result, we assessed fewer local predisposing factors in comparison to the other two studies, namely: trauma, orthodontics, extraction of adjacent tooth, irregular forces (bruxism and eruption disorders) and restoration. In the other two cohort studies (3,6) "orthodontic treatment", followed by "history of trauma" were the two most frequent observed local predisposing factors. In our study, "history of trauma" and "extraction of adjacent tooth" were the predominant ones. Although in our ECRR group, "restoration" was present with a high frequency, it was even more often found in the control group. It is of interest that in our project the number of patients with orthodontic treatment was lower for the ECRR group (n=3) than for the control group (n=5). Although the low numbers of previous orthodontic treatment limit the possibility to draw far reaching conclusions, the 3.1% frequency of orthodontic treatment as a single potential local predisposing factor in the ECRR group is in contrast with the other two cohort studies, where orthodontics overcomes the other factors with frequencies of 21.2% (3) and 45.7% (6) respectively. There are three potential explanations for this notable difference: smaller sample size, the retrospective approach in our study, which could lead to missed information, and the aforementioned difference in the age distribution. For instance, 35.7% of the patients in our study were 60-89 years old at the time of the ECRR diagnosis.

Overall, the number of several predisposing factors simultaneously present was different in the present and two earlier studies (3,6). In our study no predisposing factors were detected in 21.0% of the patients. In Heithersay's study the corresponding frequency was 15.3% (3) and in the study by Mavridou et al. 1.0% (6). An explanation for this difference, could be the number of the local factors examined, which was the lowest in our study and the highest in the Mavridou et al. study. However, there is an agreement between our findings (48.0%) and Mavridou et al. (40.0%) for the presence of a single factor as a determinant of ECRR. More than one factor was present in 28.9% (3), 31.0% (our study), and 59.0% (6) of the ECRR patients. It is noteworthy that in our study, systemic conditions (e.g. viral infections such as herpes) were not included in the analysis of the potential predisposing factors, in contrast with the study of Mavridou et al. (6).

5.2 Novelty and significance

There are no previous cohort studies about the association between systemic determinants and external cervical root resorption (ECRR). In the past, two cohort studies (3,6) have assessed the role of local predisposing factors. The present study suggests that diabetes may be a risk factor for ECRR. This indication is of significance because it may have an impact in early diagnosis of ECRR.

Cohort studies have not been conducted regarding how ECRR treatment-related determinants may be associated with the treatment outcome. The present study showed no statistically significant differences between teeth with and without ECRR treatment. This finding may indicate the potential importance of "no treatment", as a valid treatment option for some ECRR cases. Also, the presence of root canal treatment and the external cervical root resorption

83

repair combined with RCT were found to be associated with lower failure rates. These findings are relevant because they may provide some indications for better treatment planning.

Cohort studies have not been conducted regarding how local determinants may be associated with the ECRR treatment outcome, except for one study where one local determinant was examined (159). The present study indicated that posterior teeth with ECRR present higher risk of failure than anterior teeth and that the risk of failure increases as the Heithersay class increases (i.e. Class 1 showed 0.0% and Class 4 showed 47.8% risk of failure). This is of clinical importance because it can provide indications for better treatment planning, and it also highlights the significance of early ECRR diagnosis.

5.3 Limitations

It has been previously mentioned that one of the limitations of the present study is the retrospective approach to collect information from the patients' dental charts. There are local factors (e.g. oral health) that the research team would have preferred to include into the analysis. Due to the retrospective approach, it may also be possible that the MSc candidate was not able to locate all the ECRR cases in the Romexis software, although comprehensive search strategies were used. Another limitation is the relatively small sample size. Because of this, high attention to detail should be paid to the interpretation of the statistical results. For example, 75.0% of teeth with "irregular forces" had a statistically significant high failure rate. However, this percentage represents three of only four teeth with "irregular forces". A third limitation of the study is that some of the patients were not followed up. A fourth limitation is that there may be undetected teeth with ECRR (no full mouth radiographic examination contacted) or undetected patients with ECRR in the Graduate Endodontics clinic of UBC. A fifth limitation is that Heithersay

84

classification (3) was used, a classification which provides only a two-dimensional description of the resorptive defect. Due to that limitation, during the radiographic assessment; the two examiners had to come to an agreement on how to describe the cases with "well-defined" defects "extending at least at the coronal third of the root". A sixth limitation is that not all patients had a CBCT taken of the area of interest (46 patients with and 30 patients without CBCT). As a result, we were unable to use the new three-dimensional classification (51) which has been proposed by the most recent "Colleagues for Excellence of the American Association of Endodontists" (180). Also, due to the lack of CBCT, the changes of the resorptive defects during the follow-up years could not quantitatively measured.

5.4 **Future directions**

In the future, the design of the study should be a prospective cohort study instead of a retrospective cohort study. That would provide the research team with the advantage of collecting all important information by examining the patient directly. Also, a multicenter cohort study would provide a bigger sample size. In addition, use of CBCT could provide a more reliable, three-dimensional evaluation for analysis.

Chapter 6: Conclusions

- 1. Diabetes may be a potential systemic predisposing factor for external cervical root resorption.
- 2. Two treatment-related determinants are associated with lower failure rates:
- External cervical root resorption repair combined with root canal treatment
- Root canal treatment
- 3. Two local determinants are associated with higher failure rates:
- Posterior teeth
- The higher the class in the Heithersay classification is

References

- Harokopakis-Hajishengallis E. Physiologic root resorption in primary teeth: molecular and histological events. J Oral Sci. 2007;49:1–12.
- 2. AAE Glossary of Endodontic Terms. AAE. 2016;48.
- Heithersay GS. Invasive cervical resorption : an analysis of potential predisposing factors. Quintessence Int. 1999;30:83–95.
- 4. Cvek M, Lindvall AM. External root resorption following bleaching of pulpless teeth with oxygen peroxide. Dent Traumatol. 1985;1:56–60.
- 5. Heithersay GS. Invasive cervical resorption. Endodontic Topics. 2004;7:73–92.
- 6. Mavridou AM, Bergmans L, Barendregt D, Lambrechts P. Descriptive analysis of factors associated with external cervical resorption. J Endod. 2017;43:1602–10.
- Heithersay GS. Treatment of invasive cervical resorption: an analysis of results using topical application of trichloracetic acid, curettage, and restoration. Quintessence Int. 1999;30:96–110.
- 8. Hopewell-Smith A. The process of osteolysis and odontolysis, or so-called "absorption" of calcified tissues; a new- original investigation. Dent Cosm. 1930;72:323-44.
- 9. Fox J. The history and treatment of the diseases of the teeth, the gums and the alveolar processe. Swann, editor. London. 1806:122-25.
- 10. Tomes J. Dental physiology and surgery. John W. Parker, editor. London.1848:249.
- Harding, G H. The process of absorp- tion in bone and tooth structure. Br J Dent Sci. 1878;21:308–15.

- 12. Colyer, J F. Dental surgery and pathology. Longmans G& C, editor. New York. 1919:98.
- 13. Stafne EC SC. Idiopathic resorption of teeth. Am J Orthod Oral Surg. 1944;30:41–9.
- Goldman. H M. Spontaneous intermittant resorption of the teeth. J Am Dent A.ssociation. 1954;49:522–32.
- 15. Seward, G R. Periodontal disease and resorption of teeth. Br Dent J. 1963;34:443–49.
- 16. Andreasen JO. Luxation of permanent teeth due to trauma. Scan J Dent Res. 1970;78:273–
 86.
- Tronstad L. Root resorption etiology, terminology and clinical manifestations. Endod Dent Traumatol. 1988;4:241–52.
- Gunraj MN. Dental root resorption. Oral Surg Oral Med Oral Pathol Oral Radiol Endod.
 1999;88:647–53.
- 19. Ne RF, Witherspoon DE GJ. Tooth resorption. Quintessence Int. 1999;30:9–25.
- 20. Fuss Z, Tsesis I, Lin S. Root resorption--diagnosis, classification and treatment choices based on stimulation factors. Dent Traumatol. 2003;19:175–82.
- Lindskog S, Heithersay GS PA. Dental resorptions. Scandinavian Yearbook of Dentistry. Blackwell Munksgaard. 2006:322-4.
- 22. Heithersay GS. Management of tooth resorption. Aust Dent J Suppl. 2007;52:285–98.
- Patel S, Kanagasingam S, Pitt Ford T. External cervical resorption: a review. J Endod. 2009;35:616–25.
- 24. Pierce AM. Experimental basis for the management of dental resorption. Dent Traumatol.1989;5:255–65.
- 25. Wedenberg C, Yumita S. Evidence for an inhibitor of osteoclast attachment in dentinal matrix. Dent Traumatol. 1990;6:255–9.

- 26. Haapasalo M, Endal U. Internal inflammatory root resorption: the unknown resorption of the tooth. Endodontic Topics. 2006;14:60–79.
- Sahara N, Ashizawa Y, Nakamura K, Deguchi T, Suzuki K. Ultrastructural features of odontoclasts that resorb enamel in human deciduous teeth prior to shedding. 1998;252:215–28.
- 28. Domon T, Osanai M, Yasuda M, Seki E. Mononuclear odontoclast participation in tooth resorption : the distribution of nuclei in human odontoclasts. 1997;457:449–57.
- Pierce A, Berg JO, Lindskog S. Calcitonin as an alternative therapy in the treatment of root resorption. J Endod. 1988;14:459–64.
- Furseth R. The resorption processes of human deciduous teeth studied by light microscopy, microradiography and electron microscopy. Arch Oral Biol. 1968;13:417–31.
- Speziani C, Rivollier A, Gallois A, Coury F, Mazzorana M, Azocar O, et al. Murine dendritic cell transdifferentiation into osteoclasts is differentially regulated by innate and adaptive cytokines. Eur J Immunol. 2007;37:747–57.
- 32. Tyrovola JB, Spyropoulos MN, Makou M, Perrea D. Root resorption and the OPG/RANKL/RANK system: a mini review. J Oral Sci. 2008;50:367–76.
- Wedenberg C, Lindskog S. Evidence for a resorption inhibitor in dentin. Eur J Oral Sci. 1987;95:205–11.
- 34. Wedenberg C, Zetterqvist L. Internal resorption in human teeth-a histological, scanning electron microscopic, and enzyme histochemical study. J Endod. 1987;13:255–9.
- 35. Wedenberg C, Experimental LS, Wedenberg C, Lindskog S. Experimental internal resorption in monkey teeth. 1985;1:221–7.
- 36. Shaw DR, Jr FMG. Phagocytosis requires repeated triggeering of macrophage phagocytic

receptors during particle ingestion. 1981;289:401–11.

- Andreasen JO, Hjorting-Hansen E. Replantation of teeth (II). Histological study of 22 replantanted anterior teeth in human. Acta Odontol Scand. 1966;24:287-306.
- 38. Gold SI, Hasselgren G. Peripheral inflammatory root resorption: a review of the literature with case reports. J Clin Periodontol. 1992;19:523–34.
- 39. Hsien H-C, Cheng Y-A, Lee Y-L, Lan W-H, Lin C-P. Repair of perforating internal resorption with mineral trioxide aggregate: a case report. J Endod. 2003;29:538–9.
- Iqbal MK. Clinical and scanning electron microscopic features of invasive cervical resorption in a maxillary molar. Oral Surgery, Oral Med Oral Pathol Oral Radiol Endodontology. 2007;103:49–54.
- 41. Fish EW. Benign neoplasia of tooth and bone. Proc R Soc Med. 1941;34:427–32.
- 42. Wade AB. Basic Periodontology. 1963;49:64-5.
- Southam, J C. Clinical and histological aspects of peripheral cervical resorption. J Periodontoiogv. 1967;38:534–38.
- Lepp J. Roantgenographic oddities. Progressive intradental resorption. Oral Surg. 1969;27:184–5.
- 45. Makkes P.C. T van VSR. Cervical external root resorption. J Dent. 1975;3:217–22.
- 46. Cvek M. Endodontic treatment of traumatised teeth. 2nd ed. Copenhagen: Munskgaard.1981:362-3.
- Frank AL. External-internal progressive resorption and its nonsurgical correction. J Endod. 1981;710:473–6.
- Frank AL, Bakland LK. Nonendodontic therapy for supraosseous extracanal invasive resorption. J Endod. 1987;13:348–55.

- Frank AL, Torabinejad M. Diagnosis and treatment of extracanal invasive resorption. J Endod. 1998;24:500–4.
- 50. AAE and AAOMR joint position statement. Use of cone-beam-computed tomography in endodontics. Pa Dent J. 2015;78:37–9.
- 51. Patel S, Foschi F, Mannocci F, Patel K. External cervical resorption: a three-dimensional classification. Int Endod J. 2018;51:206–14.
- Goodell KB. Impact of cone-beam computed tomography on treatment planning for external cervical resorption and a novel axial slice – based classification system. J Endod. 2018;44:239–44.
- 53. Mavridou AM, Hauben E, Wevers M, Schepers E, Bergmans L, Lambrechts P.Understanding external cervical resorption in vital teeth. J Endod. 2016;42:1737–51.
- 54. Mavridou AM, Hauben E, Wevers M, Schepers E, Bergmans L, Lambrechts P. Understanding external cervical resorption patterns in endodontically treated teeth. Int Endod J. 2017;50:1116–33.
- 55. Karring T, Nyman S, Lindhe J. Healing following implantation of periodontitis affected roots into bone tissue. Journal of Clinical Periodontology. 1980;7:96–105.
- 56. Lin HJ, Chan CP, Yang CY, Wu CT, Tsai YL, Huang CC, et al. Cemental tear: clinical characteristics and its predisposing factors. J Endod. 2011;37:611–8.
- 57. Nyman S, Houston F, Sarhed G, Lindhe J, Karring T. Healing following reimplantation of teeth subjected to root planing and citric acid treatment. J Clin Periodontol. 1985;12:294–305.
- 58. Takada K, Kajiya H, Fukushima H, Okamoto F, Motokawa W, Okabe K. Calcitonin in human odontoclasts regulates root resorption activity via protein kinase A. J Bone Miner

91

Metab. 2004;22:12-8.

- Bergmans L, Van Cleynenbreugel J, Verbeken E, Wevers M, Van Meerbeek B, Lambrechts P. Cervical external root resorption in vital teeth. J Clin Periodontol. 2002;29:580–5.
- 60. Rombouts C, Giraud T, Jeanneau C, About I. Pulp vascularization during tooth development, regeneration, and therapy. J Dent Res. 2016;96:137–44.
- Hu X, Ling J, Gao Y. Effects of irrigation solutions on dentin wettability and roughness. J Endod. 2010;36:1064–7.
- Neuvald L, Consolaro A. Cementoenamel junction: microscopic analysis and external cervical resorption. J Endod. 2000;26:503–8.
- Cwyk F, Saint-pierre F TL. Endodontic implications of orthodontic tooth movement. J Dent Res. 1984;63:1039-41.
- 64. Thönen A, Peltomäki T, Patcas R, Zehnder M. Occurrence of cervical invasive root resorption in first and second molar teeth of orthodontic patients eight years after bracket removal. J Endod. 2013;39:27–30.
- Chan E, Darendeliler MA. Physical properties of root cementum: part 7. Extent of root resorption under areas of compression and tension. Am J Orthod Dentofac Orthop. 2006;129:504–10.
- 66. Dudic A, Giannopoulou C, Meda P, Montet X, Kiliaridis S. Orthodontically induced cervical root resorption in humans is associated with the amount of tooth movement. Eur J Orthod. 2017;39:534–40.
- 67. Owman-Moll P. Orthodontic tooth movement and root resorption with special reference to force magnitude and duration. A clinical and histological investigation in adolescents.
Swed Dent J. 1995;105:1–45.

- 68. Becker A, Abramovitz I, Chaushu S. Failure of treatment of impacted canines associated with invasive cervical root resorption. Angle Orthod. 2013;83:870–6.
- Yoshpe M, Kaufman A, Lin S, Gabay E, Einy S. Invasive cervical resorption following orthodontic treatment: two cases involving the same patient. Quintessence Int. 2016;47:877–84.
- 70. Heithersay GS. Invasive cervical resorption following trauma. Aust Endod J. 1999;25:79–
 85.
- 71. Trope M. Root Resorption due to Dental Trauma. Endodontic Topics. 2002;1:79–100.
- 72. Soares AJ, Souza GA, Pereira AC, Vargas-neto J, Zaia AA, Silva EJNL. Frequency of root resorption following trauma to permanent teeth. J Oral Sci. 2015;57:73–8.
- Alves T, Soares T, Barreto S, Fried H, Pereira G, Maia L, et al. Multidisciplinary approach for the treatment of extensive external cervical resorption after dental trauma. Oper Dent. 2013;38:349–57.
- 74. Grossman IL. Endodontic Practice, ed. 8, Philadelphia, Lea & Febiger, Publishers.1974:336-44.
- Harrington GW, Natkin E. External resorption associated with bleaching of pulpless teeth. J Endod. 1979;5:344–8.
- Mjor IA. Histology of the human tooth. Munksgaard, Scandinavian University Books. 1973:112–3.
- T. Lado EA, Stanley HR, Weisman MI. Cervical resorption in bleached teeth. Oral Surgery, Oral Med Oral Pathol. 1983;55:78–80.
- 78. Montgomery S. External cervical resorption after bleaching a pulpless tooth. Oral Surgery,

Oral Med Oral Pathol. 1984;57:203–6.

- 79. Friedman S, Rotstein I, Libfeld H, Stabholz A, Heling I. Incidence of external root resorption and esthetic results in 58 bleached pulpless teeth. Dent Traumatol. 1988;4:23–6.
- Gimlin DR, Schindler WG. The management of postbleaching cervical resorption. J Endod. 1990;16:292–7.
- Smith JJ, Cunningham CJ, Montgomery S. Cervical canal leakage after internal bleaching procedures. J Endod. 1992;18:476–81.
- Steiner DR, West JD. A method to determine the location and shape of an intracoronal bleach barrier. J Endod. 1994;20:304–6.
- 83. Reston E, Bueno R, Closs L, Zettermann J. Fifteen-year clinical follow-up of restoration of extensive cervical resorption in a maxillary central incisor. Oper Dent. 2017;42:55–8.
- 84. https://www.pattersondental.com
- Jiménez-Rubio A, Segura JJ. The effect of the bleaching agent sodium perborate on macrophage adhesion in vitro: implications in external cervical root resorption. J Endod. 1998;24:229–32.
- Velloso GR, de Freitas MM, Alves ATNN, Silva ARS, Barboza ESP, Moraschini V. Multiple external cervical root resorptions after home whitening treatment: a case report. Aust Dent J. 2017;62:528–33.
- 87. Vossoughi R, Takei HH. External cervical resorption associated with traumatic occlusion and pyogenic granuloma. J Can Dent Assoc. 2007;73:625–8.
- Patel K, Darbar UR, Gulabivala K. External cervical resorption associated with localized gingival overgrowth. Int Endod J. 2002;35:395–402.

- 89. Roig M, Morelló S, Mercad M, Durán-Sindreu F. Invasive cervical resorption: Report on two cases. Oral Surgery, Oral Med Oral Pathol Oral Radiol Endodontology. 2010;110:64–
 9.
- 90. Nikolidakis D, Nikou G, Meijer GJ, Jansen J a. Cervical external root resorption: 3-year follow-up of a case. J Oral Sci. 2008;50:487–91.
- Krishnan U, Moule AJ, Alawadhi A. Cone beam CT assisted re-treatment of class 3 invasive cervical resorption. BMJ Case Rep. 2015;10:1–6.
- 92. Nedir R, Lombardi T, Samsom J. Recurrent peripheral giant cell granuloma associated with cervical resorption. J Periodontol. 1997;68:381–4.
- Beertsen W, Piscaer M, Jan A, Winkelhoff V, Everts V. Generalized cervical root resorption associated with periodontal disease. J Clin Periodontol J Clin Periodontol C Munksgaard. 2001;28:1067–73.
- 94. Gunst V, Mavridou A, Huybrechts B, Van Gorp G, Bergmans L, Lambrechts P. External cervical resorption: an analysis using cone beam and microfocus computed tomography and scanning electron microscopy. Int Endod J. 2013;46:877–87.
- 95. Gunst V, Huybrechts B, De Almeida Neves A, Bergmans L, Van Meerbeek B, Lambrechts P. Playing wind instruments as a potential aetiologic cofactor in external cervical resorption: two case reports. Int Endod J. 2011;44:268–82.
- 96. O'Mahony A, McNamara C, Ireland A, Sandy J, Puryer J. Invasive cervical resorption and the oro-facial cleft patient: a review and case series. Br Dent J. 2017;222:677–81.
- 97. Patel S, Saberi N. External cervical resorption associated with the use of bisphosphonates: a case series. J Endod. 2015;41:742–8.
- 98. Solomon CS, Coffiner MO, Chalfin HE. Herpes zoster revisited: Implicated in root

resorption. J Endod. 1986;12:210–3.

- Ramchandani PL, Mellor TK. Herpes zoster associated with tooth resorption and periapical lesions. Br J Oral Maxillofac Surg. 2007;45:71–3.
- 100. Talebzadeh B, Rahimi S, Abdollahi AA, Nouroloyuni A, Asghari V. Varicella zoster virus and internal root resorption: a case report. J Endod. 2015;41:1375–81.
- Patel K, Schirru E, Niazi S, Mitchell P, Mannocci F. Multiple apical radiolucencies and external cervical resorption associated with varicella zoster virus: a case report. J Endod. 2016;42:978–83.
- 102. Jakovljevic A, Kuzmanovic Pficer J, Dragan IF, Knezevic A, Miletic M, Beljic-Ivanovic K, et al. The role of varicella zoster virus in the development of periapical pathoses and root resorption: a systematic review. J Endod. 2017;43:1230–6.
- 103. Van Wessum R, Harvey CE HP. Feline dental resorptive lesions. Prevalence patterns. Vet Clin North Am Small Anim Pract. 1992;22:1405–16.
- 104. Berger M, Schawalder P, Stich H LA. Feline dental resorptive lesions in captive and wild leopards and lions. Vet Clin North Am Small Anim Pract. 1996;13:13–21.
- 105. von Arx T, Schawalder P, Ackermann M, Bosshardt DD. Human and feline invasive cervical resorptions: the missing link?-presentation of four cases. J Endod. 2009;35:904–13.
- 106. Kumar V, Chawla A, Kaur A. Multiple Idiopathic Cervical Root Resorptions in Patients with Hepatitis B Virus Infection. J Endod. 2018;44:17–9
- Arroyo-Bote S, Bucchi C, Manzanares MC. External cervical resorption: a new oral manifestation of systemic sclerosis. J Endod. 2017;43:1740–3.
- 108. Park JK, Fava A, Carrino J, Del Grande F, Rosen A, Boin F. Association of acroosteolysis

with enhanced osteoclastogenesis and higher blood levels of vascular endothelial growth factor in systemic sclerosis. Arthritis Rheumatol. 2016;68:201–9.

- 109. Siao-Pin S, Damian LO, Muntean LM, Rednic S. Acroosteolysis in systemic sclerosis: an insight into hypoxia-related pathogenesis a review. Exp Ther Med. 2016;12:3459–63.
- 110. Neely, Anthony L; Thumbigere-Math, Vivek; Somerman, Martha J; Foster BL. A familial pattern of multiple idiopathic cervical root resorption with a 30-year follow-up. J
 Periodontol. 2017;87:1–16.
- 111. George DI Jr MR. Idiopathic resorption of teeth. A report of three cases. Am J Orthod.1986;89:13–20.
- 112. Sullivan HR JM. Idiopathic resorption. Aust Dent J. 1957;2:193–9.
- Aggarwal A, Vengal M, Ahsan A, Pai KM. Idiopathic cervical resorption: a diagnostic dilemma. Dent Update. 2007;34:646–8.
- 114. Mueller E RH. Laboratory studies of an unusual case of resorption. J Am Dent Assoc.1930;17:326–34.
- 115. Liang H, Burkes EJ FN. Multiple idiopathic cervical root resorption: systematic review and report of four cases. Dentomaxillofacial Radiol. 2003;32:150–5.
- 116. Iwamatsu-Kobayashi Y, Sato-Kuriwada S YT et al. A case of multiple idiopathic external root resorption: a 6-year follow-up study. Oral Surgery, Oral Med Oral Pathol Oral Radiol Endod. 2005;100:772–9.
- 117. Yu VSH, Messer HH, Tan KB. Multiple idiopathic cervical resorption: case report and discussion of management options. Int Endod J. 2011;44:77–85.
- 118. Wu J, Lin LY, Yang J, Chen XF, Ge JY, Wu JR, et al. Multiple idiopathic cervical root resorption: a case report. Int Endod J. 2016;49:189–202.

- 119. Coyle M, Toner M, Barry H. Multiple teeth showing invasive cervical resorption an entity with little known histologic features. J Oral Pathol Med. 2006;35:55–7.
- 120. Prakash S, Kesary R, Mengji AK, Yaga US, Panta P. An unusual case of idiopathic multiple invasive cervical resorption. 2014;26:5–7.
- MacDonald-Jankowski D, Liang H, Burkes EJ, Frederiksen NL. Multiple idiopathic cervical root resorption most frequently seen in younger females. Evid Based Dent. 2005;6:20-4.
- Cholia SS, Wilson PH MJ. Multiple idiopathic external apical root resorption: report of four cases. Dentomaxillofacial Radiol. 2005;34:240–6.
- 123. Kalender A, Oztan MD, Basmaci F, Aksoy U, Orhan K. CBCT evaluation of multiple idiopathic internal resorptions in permanent molars: case report. BMC Oral Health. 2014;14:39-45.
- Schwartz RS, Robbins JW, Rindler E. Management of invasive cervical resorption: observations from three private practices and a report of three cases. J Endod. 2010;36:1721–30.
- 125. Shemesh A, Ben Itzhak J, Solomonov M. Minimally invasive treatment of class 4 invasive cervical resorption with internal approach: a case series. J Endod. 2017;43:1901–8.
- 126. Kqiku L, Ebeleseder K, Glockner K. Treatment of invasive cervical resorption with sandwich technique using mineral trioxide aggregate: a case report. Oper Dent. 2012;37:98–106.
- 127. Patel S, Dawood A. The use of cone beam computed tomography in the management of external cervical resorption lesions. Int Endod J. 2007;40:730–7.
- 128. Dias C, Closs L, Barletta F, Reston E, Tovo MF, Lambert P. Root resorption a 6-year

follow-up case report. 2015;9:103–5.

- 129. Gulsahi A, Gulsahi K, Ungor M. Invasive cervical resorption: clinical and radiological diagnosis and treatment of 3 cases. Oral Surgery, Oral Med Oral Pathol Oral Radiol Endodontology. 2007;103:65–72.
- 130. Karypidou A, Chatzinikolaou I-D, Kouros P, Koulaouzidou E, Economides N.
 Management of bilateral invasive cervical resorption lesions in maxillary incisors using a novel calcium silicate-based cement: a case report. Quintessence Int. 2016;47:637–42.
- Vasconcelos KDF, De-Azevedo-Vaz SL, Freitas DQ, Haiter-Neto F. CBCT postprocessing tools to manage the progression of invasive cervical resorption: a case report. Braz Dent J. 2016;27:476–80.
- Salzano S, Tirone F. Conservative nonsurgical treatment of class 4 invasive cervical resorption: a case series. J Endod. 2015;41:1907–12.
- 133. Smidt A, Nuni E, Keinan D. Invasive cervical root resorption: treatment rationale with an interdisciplinary approach. J Endod. 2007;33:1383–7.
- 134. Asgary S, Fazlyab M. Surgical repair of invasive cervical root resorption with calciumenriched mixture cement: a case report. Gen Dent. 2015;63:37–40.
- Hiremath H, Yakub SS, Metgud S, Bhagwat S V., Kulkarni S. Invasive Cervical resorption: a case report. J Endod. 2007;33:999–1003.
- Libonati A, Montella D, Montemurro E, Campanella V. External cervical resorption: a case report. Eur J Paediatr Dent. 2017;18:296–8.
- 137. Singla M. External cervical resorption a diagnostic challenge. 2012;8:199–203.
- 138. Tsaousoglou P, Markou E, Efthimiades N, Vouros I. Characteristics and treatment of invasive cervical resorption in vital teeth. A narrative review and a report of two cases. Br

Dent J. 2017;222:423-8.

- 139. Silveira FF, Nunes E, Soares JA, Ferreira CL, Rotstein I. Double "pink tooth" associated with extensive internal root resorption after orthodontic treatment: a case report. Dent Traumatol. 2009;25:43–7.
- 140. Patel S, Dawood A, Wilson R, Horner K, Mannocci F. The detection and management of root resorption lesions using intraoral radiography and cone beam computed tomography an in vivo investigation. Int Endod J. 2009;42:831–8.
- Rahmatulla, M; Wyne AH. Classification of cervical burnout and its distribution in the dentition. Indian J Dent Res. 1995;6:13–9.
- 142. Andreasen FM, Sewerin I, Mandel U, Andreasen JO. Radiographic assessment of simulated root resorption cavities. Dent Traumatol. 1987;3:21–7.
- 143. Kamburoğlu K, Kurşun Ş, Yüksel S, Öztaş B. Observer ability to detect ex vivo simulated internal or external cervical root resorption. J Endod. 2011;37:168–75.
- 144. Bernardes RA, de Paulo RS, Pereira LO, Duarte MAH, Ordinola-Zapata R, de Azevedo JR. Comparative study of cone beam computed tomography and intraoral periapical radiographs in diagnosis of lingual-simulated external root resorptions. Dent Traumatol. 2012;28:268–72.
- 145. Vaz de Souza D, Schirru E, Mannocci F, Foschi F, Patel S. External cervical resorption: a comparison of the diagnostic efficacy using 2 different cone-beam computed tomographic units and periapical radiographs. J Endod. 2017;43:121–5.
- 146. AAE Cone Beam-Computed Tomography in Endodontics. Colleages Excell. 2011;1–7.
- 147. Patel S, Durack C, Abella F, Shemesh H, Roig M, Lemberg K. Cone beam computed tomography in endodontics a review. Int Endod J. 2015;48:3–15.

- 148. Patel K, Mannocci F, Patel S. The assessment and management of external cervical resorption with periapical radiographs and cone-beam computed tomography: a clinical study. J Endod. 2016;42:1435–40.
- 149. Mavridou AM, Pyka G, Kerckhofs G, Wevers M, Bergmans L, Gunst V, et al. A novel multimodular methodology to investigate external cervical tooth resorption. Int Endod J. 2016;49:287–300.
- 150. Estevez R, Aranguren J, Escorial A, De Gregorio C, De La Torre F, Vera J, et al. Invasive cervical resorption class III in a maxillary central incisor: diagnosis and follow-up by means of cone-beam computed tomography. J Endod. 2010;36:2012–4.
- 151. Fernández R, Rincón JG. Surgical endodontic management of an invasive cervical resorption class 4 with mineral trioxide aggregate: a 6-year follow-up. Oral Surgery, Oral Med Oral Pathol Oral Radiol Endodontology. 2011;112:18–22.
- 152. Harris BT, Caicedo R, Lin WS, Morton D. Treatment of a maxillary central incisor with class III invasive cervical resorption and compromised ferrule: a clinical report. J Prosthet Dent. 2014;111:356–61.
- 153. Pruthi PJ, Dharmani U, Roongta R, Talwar S. Management of external perforating root resorption by intentional replantation followed by biodentine restoration. Dent Res J . 2015;12:488–93.
- 154. Frank AL. Extracanal invasive resorption: an update. Compend Contin Educ Dent.1995;3:250–66.
- 155. Haapasalo M, Parhar M, Huang X, Wei X, Lin J, Shen Y. Clinical use of bioceramic materials. Endodontic Topics. 2015;32:97–117.
- 156. Rankow HJ, Krasner PR. Endodontic applications of guided tissue regeneration in

endodontic surgery. J Endod. 1996;22:34-43.

- 157. AAE Treatment Options for the Compromised Tooth. AAE. 2011;1–16.
- 158. Nasser AZ, Mahdian M, Fedorowicz M, Ma M, Ahangari Z, Nasser M, et al. Interventions for the management of external root resorption a review. Cochrane Database Syst Rev. 2015;2:1-16.
- Heithersay GS. Treatment of invasive cervical resorption: an analysis of results using topical application of trichioracetic acid, curettage, and restoration. Quintessence Int. 1999;30:96-110.
- 160. Lo Giudice G, Matarese G, Lizio A, Lo Giudice R, Tumedei M, Zizzari V, et al. Invasive cervical resorption: a case series with 3-year follow-up. Int J Periodontics Restorative Dent. 2016;36:103–9.
- 161. Asgary S, Nosrat A. Conservative Management of class 4 invasive cervical root resorption using calcium-enriched mixture cement. J Endod. 2016;42:1291–4.
- Meister F, Haasch GC, Gerstein H. Treatment of external resorption by a combined endodontic-periodontic procedure. J Endod. 1986;12:542–5.
- 163. Chang YC, Lin HJ, Lee YL, Liu LC, Jeng JH. Repairing invasive cervical root resorption by glass ionomer cement combined with mineral trioxide aggregate. J Dent Sci. 2012;7:395–9.
- 164. Vijetha B, Rangappa J, Reddy S, Aspalli N. Management of external cervical resorption using mineral trioxide aggregate. Indian J Oral. 2012;3:161-4
- 165. Park JB, Lee JH. Use of mineral trioxide aggregrate in the non-surgical repair of perforating invasive cervical resorption. Med Oral Patol Oral Cir Bucal. 2008;13:678–80.
- 166. Yilmaz HG, Kalender A, Cengiz E. Use of mineral trioxide aggregate in the treatment of

invasive cervical resorption: a case report. J Endod. 2010;36:160–3.

- 167. Tavares WLF, Lopes RCP, Oliveira RR, Souza RG de, Henriques LCF, Ribeiro-Sobrinho AP. Surgical management of invasive cervical resorption using resin-modified glass ionomer cement. Gen Dent. 2013;61:16-8.
- 168. Vinothkumar TS, Tamilselvi R, Kandaswamy D. Reverse sandwich restoration for the management of invasive cervical resorption: a case report. J Endod. 2011;37:706–10.
- 169. Pace R, Giuliani V, Pagavino G. Mineral trioxide aggregate in the treatment of external invasive resorption: a case report. Int Endod J. 2008;41:258–66.
- 170. Martos J, César Neto J, Silveira LF, Silveira C, Piovesan E. Clinical technique for invasive cervical root resorption. J Conserv Dent. 2011;14:440-4.
- 171. Baratto-Filho F, Limongi O, Araújo C de JK, Neto MDS, Maia SMAS, Albuquerque DS. Treatment of invasive cervical resorption with MTA: case report. Aust Endod J. 2005;31:76–80.
- 172. Haapasalo M. Level of evidence in endodontics: what does it mean? Endodontic Topics.2016;34:30–41.
- 173. Reit G. Decision strategies in endodontics: on the design of a recall program. Dent Traumatol. 1987;3:233–9.
- 174. Ørstavik D. Time-course and risk analyses of the development and healing of chronic apical periodontitis in man. Int Endod J. 1996;29:150–5.
- Friedman S, Mor C. The success of endodontic therapy-healing and functionality. J Calif Dent Assoc. 2004;32:493–503.
- 176. Friedman S, Abitbol S, Lawrence HP. Treatment outcome in endodontics : the toronto study . phase 1 : initial treatment. 2003;29:787-93.

- 177. de Chevigny C, Dao TT, Basrani BR, Marquis V, Farzaneh M, Abitbol S, et al. Treatment outcome in endodontics: the toronto study—phases 3 and 4: orthograde retreatment. J Endod. 2008;34:131–7.
- 178. Norouzirad R, González-Muniesa P, Ghasemi A. Hypoxia in obesity and diabetes:
 potential therapeutic effects of hyperoxia and nitrate. Oxid Med Cell Longev. 2017;20:31-45.
- 179. Kang SH, Kim BS, Kim Y. Cracked teeth: Distribution, characteristics, and survival after root canal treatment. J Endod. 2016;42:557–62.
- 180. AAE The impact of cone beam computed tomography in endodontics: a new era in diagnosis and treatment planning. Colleages Excell. 2018;1–7.